# COLLECTIVE AGREEMENT

Between

# TRENT UNIVERSITY



And

# ONTARIO PUBLIC SERVICE EMPLOYEES UNION

LOCAL 365



JULY 1, 2018 TO JUNE 30, 2022

ARTICLE	1 – PURPOSE	8
1.1	General Purpose	8
1.2	Application	8
ARTICLE	2 - NO STRIKES OR LOCKOUTS	9
ARTICLE	3 - MANAGEMENT RIGHTS	10
3.1	Functions and Prerogatives	10
3.2	Rights	10
3.3	Exercise of Rights	10
ARTICLE	4 - UNION BUSINESS	11
4.1	Recognition	11
4.2	No Interference	11
4.3	Union Function	12
4.4	OPSEU Membership and Dues	12
4.5	OPSEU Information from Department of Human Resources	13
4.6	Union Activities	13
4.7	Stewards	13
4.8	Rights of Stewards	13
4.9	Release and Assistance to Officers and Stewards	14
4.10	Release Time for OPSEU Officers	15
4.11	Members' Participation in Union Functions	15
4.12	Election to Provincial Executive Board	16
4.13	Negotiating Committee	17
4.14	New Staff Interviews	17
4.15	Staff Interviews	17
ARTICLE	5 - HUMAN RIGHTS AND EMPLOYMENT EQUITY	18
5.1	Discrimination and Harassment	18
5.2	Employment Equity	18
ARTICLE	6 - ADMINISTRATION OF THE AGREEMENT	20
6.1	Relations Between the Parties	20
6.2	Advisors and Representatives	20
6.3	Joint Committee on the Administration of the Agreement	20
6.4	Addresses for Official Communications	21
6.5	Term of Agreement	21
6.6	Renegotiation of Agreement	21
6.7	Copies of the Agreement	22

ARTICLE	7 - HEALTH AND SAFETY	. 23
7.1	General Statement	. 23
7.2	Health and Safety Committee	. 23
7.3	First Aid	. 23
ARTICLE	8 - STAFF INFORMATION	. 25
8.1	Personnel Records	. 25
8.2	Confidentiality of Personnel Files	. 25
ARTICLE	9 - DISCIPLINE AND DISCHARGE	. 26
9.1	University Responsibility	. 26
9.2	Discipline Policy	. 26
9.3	OPSEU Representation	. 26
9.4	Notification	. 26
ARTICLE	10 - EMPLOYMENT	. 27
10.1	Position Definitions	. 27
10.2	Changes in Position Definition	. 30
10.3	Conversions to Regular or Recurring Positions	. 30
10.4	Job Descriptions and Classifications	. 30
10.5	Probationary Period	. 30
10.6	Medical Examinations	. 31
10.7	Seniority	. 31
10.8	Termination	. 32
10.9	Retirement	. 32
10.10	Resignation	. 33
10.11	Failure to Satisfactorily Complete the Probationary Period	. 33
10.12	Termination for Cause	. 33
10.13	Unsatisfactory Work as Just Cause for Termination	. 33
10.14	Illness or Disability	. 34
10.15	Death	. 34
10.16	Completion of Contracted Employment	. 35
10.17	Severance Pay on Termination	. 35
10.18	Technological Change	. 35
10.19	Compensation During Retraining	. 36
10.20	Salary Compensation for New Position	. 36
10.21	Alternate Employment	. 36
ARTICLE	11 - REDUNDANCY AND LAYOFF	. 37
11.1	Contracting Out	. 37
11.2	Contracting In	. 37

11.3	New Work	
11.4	Layoff	
11.5	Notice and Severance	
11.6	Redeployment/Displacement	
11.7	Recall Rights	
11.8	Emergency Temporary Layoffs	41
11.9	Labour Disputes	41
ARTICLE	12 - HOURS OF WORK	
12.1	Standard Work Week	
12.2	Modified Work Week	
12.3	Overtime	
12.4	Lunch Period	
12.5	Rest Periods	
12.6	Scheduled On-Call	
12.7	Working During Emergency Closure	
ARTICLE	13 - COMPENSATION	
13.1	Pay Equity	
13.2	Salary Grid and Salary Bands * [*See S	Schedule A] 45
13.3	Overtime	
13.3 13.4	Overtime Emergency Call-in	
		45
13.4	Emergency Call-in	45 46
13.4 13.5	Emergency Call-in Supplementary Payment	
13.4 13.5 13.6	Emergency Call-in Supplementary Payment Salary Reductions	
13.4 13.5 13.6 13.7	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid	45 46 46 46 46 47
13.4 13.5 13.6 13.7 13.8	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases	45 46 46 46 47 47
13.4 13.5 13.6 13.7 13.8 13.9	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates	45 46 46 46 47 47 47
13.4 13.5 13.6 13.7 13.8 13.9 13.10	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions.	45 46 46 46 47 47 47 47 47 48 49
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions Assignments Outside the Bargaining Unit	45 46 46 46 47 47 47 47 47 48 49 49
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions.	45 46 46 46 47 47 47 47 47 48 49 49
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions Assignments Outside the Bargaining Unit	45 46 46 46 47 47 47 47 47 47 49 49 49
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions Assignments Outside the Bargaining Unit Temporary Assignments	45 46 46 46 47 47 47 47 47 47 48 49 49 49 50
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions Assignments Outside the Bargaining Unit Temporary Assignments Performance Appraisal	45 46 46 47 47 47 47 47 48 49 49 49 50 51
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15 13.16 13.17 13.18	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions Assignments Outside the Bargaining Unit Temporary Assignments Performance Appraisal Payroll Matters Payroll Deductions Garnishees, Assignments, Third-Party Demands	45 46 46 46 47 47 47 47 47 48 49 49 49 49 50 50 51 52 52
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15 13.16 13.17 13.18	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions Assignments Outside the Bargaining Unit Temporary Assignments Performance Appraisal Payroll Matters Payroll Deductions	45 46 46 46 47 47 47 47 47 47 49 49 49 49 50 50 51 52 52
13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15 13.16 13.17 13.18	Emergency Call-in Supplementary Payment Salary Reductions Ongoing Administration of the Salary Grid Step Increases Annual/Hourly Rates Promotions Assessment and Reassignment Demotions Assignments Outside the Bargaining Unit Temporary Assignments Performance Appraisal Payroll Matters Payroll Deductions Garnishees, Assignments, Third-Party Demands	45 46 46 46 47 47 47 47 47 48 49 49 49 49 50 50 51 52 52 52 52 53

14.3	Insured Group Benefits	. 53
14.4	Trent Pension Plan	. 54
14.5	Tendering of Benefits Contract	. 55
14.6	Coverage	. 55
14.7	Coverage and Cost Sharing	. 57
14.8	Benefit Coverage Participation	. 57
14.9	Travel Expenses	. 58
14.10	Professional Association	. 58
14.11	Flexible Benefits Plan	. 58
14.12	Automobile Allowance	. 59
ARTICLE	E 15 - LEAVES OF ABSENCE WITH PAY	. 60
15.1	Vacations	. 60
15.2	Sickness/Accident/Bereavement During Vacation	. 61
15.3	Holidays	. 61
15.4	OPSEU Day (Floating Day)	. 62
15.5	Reporting Absence	. 62
15.6	Medical Certificate	. 63
15.7	Sick Leave Policy for Staff Covered Under the Long Term Disability Plan	. 63
15.8	Sick Leave Policy for Staff Not Covered by the Long Term Disability Plan	. 64
15.9 Insurai	Sick Leave Policy for Staff Covered Under the Workplace Safety and nce Board	. 64
15.10	Bereavement Leave	. 65
15.11	Civic Leave	. 65
15.12	Compassionate Leave and Special Leaves	. 66
15.13	Medical and Dental Appointments	. 66
15.14	Maternity/ Pregnancy Leave Sub Plan	. 67
15.15	Parental Leave Sub Plan	. 67
15.16	Applications	. 69
15.17	Deferred Income Leave	. 69
15.18	Long Service Award	. 69
ARTICLE	E 16 - LEAVES OF ABSENCE WITHOUT PAY	. 70
16.1	Leaves of Absence Without Pay	. 70
16.2	Pregnancy Leave	. 70
16.3	Parental Leave	. 71
16.4	Long-term Leave of Absence	.71
16.5	Short-term Leave of Absence	.71
ARTICLE	TT - JOB CLASSIFICATION	.73

17.1	Job Descriptions	73
17.2	Job Questionnaires	73
17.3	Joint Job Evaluation System	73
17.4	The Joint Job Evaluation Committee	73
17.5	Joint Working Sub-Committee	74
17.6	New Jobs	74
17.7	Substantively-altered Jobs	75
17.8	Mediation/Arbitration	75
17.9	Appeals of Classification Reviews	76
17.10	Band Increase	76
17.11	Grievance of Classification Reviews	76
17.12	Cost Sharing	78
ARTICLE	18 - JOB POSTING PROCEDURES	79
18.1	Job Postings	79
18.2	Applications	79
18.3	Selection	80
18.4	Release from Present Position	81
ARTICLE	19 - CAREER DEVELOPMENT	82
19.2	Academic Courses and Professional Study	83
19.3	Education Leave	84
19.4	Policy Respecting Rebate of Tuition Course Fees to Eligible University Stafe	
	rs	
19.5	Dependent's Tuition Waiver	
19.6	Definitions of Dependency	
19.7	On-the-Job Training	
19.8	University Committees	
	20 - GRIEVANCE AND ARBITRATION	
20.1	Grievance Definition	
20.2	Types of Grievance	
20.3	Carriage of Grievances	87
20.4	Informal Complaint Stage	
20.5	Formal Grievance Steps	
20.6	General Provisions	
20.7	Referral of Grievance to Arbitration	90
20.8	Selection of Chairperson	90
20.9	Appointment of a Single Mediator/Arbitrator	90
20.10	Exclusions	90

20.11	Jurisdiction	90
20.12	Fees and Expenses	91
20.13	Decision of the Board	91
ARTICLE	21- ANIMAL CARE WORKERS	92
APPEND	X I- Pension Health Care Allowance	93
APPEND	X II- Student Employees	95
APPEND	IX III- Pension Plan	
APPEND	IX IV- Restructuring	100
APPEND	X V- RLC – Essential Workers	101
APPEND	X VI- LTD Benefits	102
APPEND	IX VII- Lieu Time	103
APPEND	X VIII- Benefits over age 65	104
APPEND	IX IX- Training	105
APPEND	X X- Backfill of Contract Positions	106
APPEND	X XI- Job Competition Equivalency	108
	X XII- Survivor Income Benefit	
APPEND	X XIII- Pension Plan 2018-2022	110
APPEND	X XIV- Vacation pay-out for Regular Caretaking Staff	111
APPEND	X XV- Vacation for Recurring Employees	112

# ARTICLE 1 – PURPOSE

#### 1.1 <u>General Purpose</u>

The general purpose of the Agreement is to establish an orderly bargaining relationship between the Board of Governors of Trent University, hereafter known as the University, and its staff members represented by the Ontario Public Service Employees Union, hereafter known as the Union, to ensure the peaceful settlement of disputes and to set forth agreement covering rates of pay and other working conditions which shall supersede all previous agreements and arrangements between the University and the staff members represented by the Union.

#### 1.2 <u>Application</u>

This Agreement applies equally to staff members whose salaries, wages and benefits are charged to ancillary or general operating budgets. Ancillary budgets are:

Athletics Bookstore College Conferences Dining Services Health Services Print Shop Transportation and Parking

and any other similar budgets under the responsibility of the Board.

# **ARTICLE 2 - NO STRIKES OR LOCKOUTS**

The Union agrees that there shall be no strikes, and the University agrees that there shall be no lockouts during the term of this Agreement.

In the event that any employees at Trent University, other than those covered by this Agreement, engage in a lawful strike and maintain picket lines, staff members covered by this Agreement shall not be required to perform the work normally done by those employees.

# ARTICLE 3 - MANAGEMENT RIGHTS

#### 3.1 <u>Functions and Prerogatives</u>

All inherent and common law management functions and prerogatives which the University has not expressly modified or restricted by a specific provision of this Agreement are retained and vested exclusively in the University.

#### 3.2 Rights

Without limiting the generality of the above, these rights include, but are not limited to, the right to:

- (a) discharge, reprimand, suspend, or otherwise discipline staff for just cause; and to hire, classify, direct, assign duties, promote, demote, transfer, layoff or recall staff;
- (b) determine the requirements of a job and the standards of the work to be performed;
- (c) expand, reduce, alter, combine, transfer or cease any job, department, operation or service;
- (d) determine the size and composition of the work force;
- (e) make or change rules, policies and practices provided such rules, policies and practices shall not be inconsistent with the spirit or terms of this Agreement;
- (f) maintain order and efficiency and otherwise generally manage the University, direct the work force and establish terms and conditions of employment not in conflict with the provisions of this Agreement.

#### 3.3 Exercise of Rights

Management shall exercise its rights in a manner that is consistent with the terms of this Agreement.

# **ARTICLE 4 - UNION BUSINESS**

### 4.1 Recognition

The University recognizes the Ontario Public Service Employees Union (OPSEU) as the sole bargaining agent for all non-academic employees of Trent University in the City of Peterborough, and any of its locations in Ontario where bargaining unit work of the University is being performed, save and except supervisors and persons above the rank of supervisor, employees for whom any trade union held bargaining rights as of May 27, 1993, President, Vice-Presidents, Associate Vice-Presidents, Deans, Associate Deans, College Heads, research assistants, research technicians and postdoctoral fellows, students including graduate students, University Secretary and Secretary to the Board of Governors, Directors of Human Resources, Athletics, Communications, Physical Resources, Finance, Alumni Affairs, Computer Services, Hospitality and Conference Services, Assistant Directors, Registrar, Associate Registrar, Assistant Registrars, Manager Liaison Services, Assistant to the Dean of Arts and Science, Manager of the Bookstore, Manager of Facility Services, Manager Accounting Services, Telecommunications Manager, Co-ordinator of Counselling, Manager Budgeting Services, Operations Engineer, Construction Planning Coordinator, Academic Counsellor and Administrative Assistant, Secretaries to the President's Office, Assistant University Secretaries, Physical Resources Assistant, University Librarian, Executive Secretary to the VP Administration and Finance, Administrative Assistant Dean's Office, Security Guards, Director of Student Health Services, Manager Purchasing Services, Manager Audio-Visual Services, VAX Systems Manager, Institutional Analyst, Human Rights Officer, Communications Officer, College Administrators, Library Administrative Assistant, Executive Secretary to the Vice-President (University Services), Health and Safety Officer, Payroll Authorization and Administrative Secretary, Human Resources Administrative Assistant and Secretary, Payroll Supervisor, Payroll Clerk, Data Entry Clerk Human Resources, Administrative Secretary/Assistant Athletics, Administrative Secretary Dean's Office, Administrative Secretary University Secretariat, and Members of the professions of law, dentistry, architecture, medicine, engineering, land surveying, accounting and horticulture.

It is further understood and agreed that the following persons are excluded from the bargaining unit:

i) stand-by employees as defined in Article 10.1(e).

#### 4.2 No Interference

The University shall not be permitted to make any written or verbal agreement with any employee in the bargaining unit which conflicts with the terms of the Agreement signed on behalf of Trent University and OPSEU.

No employee in the bargaining unit shall be required or permitted to make a written or verbal agreement with the University or its representative(s) which

conflicts with the terms of the Agreement signed on behalf of Trent University and OPSEU.

# 4.3 Union Function

The University recognizes that it is the function of OPSEU to:

- (a) establish the right to bargain collectively, on behalf of members of the bargaining unit, their terms and conditions of employment;
- (b) provide prompt and equitable disposition of grievances; and,
- (c) regulate relations between the employer and members of the bargaining unit.

# 4.4 OPSEU Membership and Dues

The Union acknowledges that it is the right and the privilege of any staff member in the bargaining unit to become, or refrain from becoming, a member of the Union, and no employee shall be required to join the Union as a condition of employment.

The parties hereto mutually agree that any employee of the University covered by this Agreement may become a member of the Union if he/she wishes to do so, and refrain from becoming a member of the Union if he/she desires.

The University shall deduct from every employee in the bargaining unit, any monthly dues as may be designated by the Union from time to time.

Deductions for Union dues shall be made from the payroll of every pay period each month and shall be forwarded to the Accounting Department of the Union at 100 Lesmill Road, North York, Ontario, M3B 3P8, not later than the 15th day of the month following, accompanied by a list of the names and Social Insurance Numbers of all employees from whose wages the deductions have been made, and as Income Tax T-4 slips are made available, the University shall show on the said statement of income, the amount so deducted for the calendar year.

The Union agrees to indemnify and save the University harmless against all claims or other forms of liability that may arise out of, or by reason of, deductions, payments or the provision of employee information supplied by the Employer in accordance with this Article. The parties further agree that requirements in this Article concerning the provision of personal information shall be in conformity with Freedom of Information/Protection of Privacy regulations which may, in future, be imposed on the University, or adopted by the parties.

The parties agree that the current formula of Union dues deduction will be applied to retroactive wage increases.

# 4.5 OPSEU Information from Department of Human Resources

The Union shall receive annually a list containing the names of all staff members, their job title and classification.

The Union shall be advised each month of all hirings, changes in classification or type of appointment, transfers and terminations involving regular and recurring positions. The employer on a yearly basis shall provide to the union a report summarizing all compensated overtime worked by each employee in the bargaining unit. Any contractual appointment lasting longer than one month (excluding students) will be listed, including the duration of each appointment. Notwithstanding the fact that students, including graduate students, are excluded from the bargaining unit in accordance with the decision of the Ontario Labour Relations Board dated September 22, 1993, it is understood that in the event a regular or recurring employee, absent because of leave, illness or disability, is temporarily replaced by a student scheduled to work 17.5 hours or more per week for a period in excess of sixty (60) consecutive calendar days, the University will provide the above-referenced information to the Union.

### 4.6 Union Activities

The University shall provide to the Union without charge suitable meeting rooms on the University campus for the conduct of Union business, subject only to normal scheduling arrangements. The use of other facilities and services available to University departments shall be provided to the Union at rates as charged to departments.

#### 4.7 Stewards

The University shall recognize the right of the Union to appoint or otherwise select stewards, one of whom shall be the Chief Steward. The Chief Steward may act for the Union in any grievance where the steward is unavailable. The Chief Steward, or the Chief Steward's designate, shall be responsible for the initial presentation of all policy and group grievances, or any other grievances arising directly between the University and the Union.

#### 4.8 Rights of Stewards

The right of stewards or officers to leave their work without loss of pay to investigate grievances is granted on the following conditions:

- i) the steward or officer will provide the supervisor with written notice as far in advance of the absence as possible.
- ii) the steward or officer shall obtain permission of the immediate supervisor before leaving work, such permission shall not be unreasonably withheld;
- iii) the time off from work shall be devoted to the prompt handling of

grievances and shall be limited to the time required for a meeting with the grievor(s) to obtain the relevant facts surrounding the grievance (to a maximum of thirty (30) minutes per grievance, to be scheduled, outside of the grievor's/grievors' normal working hours), attending to grievance step meetings with management, and for the steward meeting with the appropriate supervisor for the purpose of discussing a possible resolution, under Article 20.4. Employees having grievances shall not discuss these with a steward during the employee's working hours, but an employee may request time during his/her working hours to accommodate such a discussion with the understanding the time will be made up as agreed with his/her supervisor. A request of this nature will not be unreasonably denied.

#### 4.9 Release and Assistance to Officers and Stewards

The University agrees that leave of absence without loss of regular salary, wages or benefits may be granted to officers and stewards of the Union to attend conferences, workshops, seminars and other meetings outside the University in connection with the affairs of the Union. The salary or wage costs associated with such leaves shall be equally shared by the parties.

The leave granted to any individual shall not be withheld unreasonably, but shall be subject to the reasonable workplace requirements. Staff members desiring such leave shall make their requests in writing with reasonable notice to their supervisor with a copy of the request to the Department of Human Resources. The combined total of such release time shall be thirty (30) days per contract year. If additional time off is required, the time may be granted without pay and with benefits.

The University recognizes the contribution of staff members who assume the responsibilities required of offices within the Union. The Union will attempt to rotate officers so that no one segment of the University is regularly depleted because of OPSEU responsibilities. The University anticipates that a certain amount of time off during normal working hours will be required for phone calls, research and meetings, which are necessary to assure the normal functioning of such an organization and the regulation of relations between the University and the staff members. A list of officers and agents shall be forwarded by the Union to the University within ten (10) working days of the appointment of such officers and agents to such offices. The University shall not be required to recognize any such person until it has received such notification.

The Department of Human Resources shall inform Department Heads affected of the names of staff members involved. Such staff members' attendance at meetings and other Union business in general shall be without prejudice to the staff members' salary/wage, benefits and other terms and conditions of employment, or to any of their rights and privileges within the University. While such availability shall not be unreasonably denied, time off shall not interfere with the regular performance of a member's duties and responsibilities to the University, nor require payment of overtime premiums. Without limiting the generality of the preceding, the amount of relief from duty required to assure the normal functioning of the Union shall be decided by the Joint Committee.

Arrangements for contractual appointment to fulfill the workload requirements, etc., of the positions held by such officers and stewards shall be made by arrangement with the Department of Human Resources, subject to the needs of the department.

The selection of any such employee(s) will be at the discretion of supervisors and/or department heads.

# 4.10 Release Time for OPSEU Officers

The University agrees, where necessary, to grant up to an aggregate maximum of seventy-two (72) days per year of release time with pay for officer(s) of the Local Union, including the President. The University will backfill the local President for 16 hours per week. Should the Union exceed the allocation of 72 days, the University will grant up to an additional 25 paid days upon request by the union. The President will discuss the need for such release time with the appropriate supervisor and with the Associate Vice President (Human Resources). Once it is agreed that release time is required, a schedule will be worked out with the supervisor, and a copy will be sent to the Associate Vice President (Human Resources). Rather than take time off, the President may use these funds to provide clerical/secretarial support for the position.

This release time is to be used by the President and/or officer(s) to attend to responsibilities flowing directly from the position of President or officer of the Union. Every effort will be made by the President and officers to schedule meetings and interviews during this period rather than during regular working hours.

#### 4.11 <u>Members' Participation in Union Functions</u>

The University acknowledges the obligation of members of the Executive, Steward Body, Nominating Committee, Grievance Committee, Communications Committee and any other committee as agreed upon by the Joint Committee of the bargaining unit to participate in the business of the Union. Subject to operating requirements and the approval of their supervisor and Department of Human Resources, members of the Executive and the above-mentioned committees shall be entitled to make reasonable adjustments of their normal duties, without loss of salary/wages or benefits, for the purpose of participation in the business of the Union. A member's service to the Union shall not be negatively regarded for purposes of consideration for promotion and merit increments.

All members in Peterborough and Oshawa will be allowed a two (2) hour lunch break each to attend each of the following membership meetings:

- (1) Annual General Membership Meeting
- (2) Collective Agreement Ratification (Limit of 2)
- (3) One (1) Special General Meeting.

Members in Oshawa shall be provided the appropriate travel time to attend the above meetings, and will provide notice to their supervisor five (5) days in advance of the meeting.

The Union agrees to provide at least ten (10) working days' notice to the University of the date of the Annual General Membership Meeting, and as much notice as possible with respect to Ratification and Special General Meetings.

#### 4.12 Election to Provincial Executive Board

Upon request of the Union, confirmed in writing, and provided that seven (7) days written notice period is given, leave of absence with no loss of pay and with no loss of credits shall be granted to employees elected as Executive Board members and Executive Officers of the Union, for the purpose of conducting the internal business affairs of the Union.

The Union will reimburse the University for the wages paid to members of the Executive Board or Executive Officers or the replacement costs if such costs are greater than the wages paid where a leave of absence is granted.

When an employee is elected as the Union's President or first Vice-President, the Union will, immediately following such election, advise the University of the name of the employee so elected. Leave of absence with pay shall be granted from the employee's place of employment for the duration of the current term of office.

During the term of such leave of absence, the Union will reimburse the University for the salary paid to the employee on such leave of absence and contribute the University's share of contributions to the pension fund and the Canada Pension Plan. The Union will make the University's contribution to any prevailing health or other plans applicable to the elected employee, during the leave of absence. The Union will make the University's contribution for Employment Insurance (EI).

On completion of the employee's term of office, the President or first Vice-President may return to their previous employment, and service shall be deemed to be continuous for all purposes. Any leave of absence extending beyond the initial term of office of the President or first Vice-President shall be a matter to be determined between the parties, and such additional leave shall be subject to the same conditions and terms as prevailed in the initial leave of absence.

It is understood that replacement staff for temporary vacancies created under the provisions of this Article may be employed in a contractual or temporary capacity for the full duration of the vacancy notwithstanding any time limits on such appointments elsewhere in the Agreement.

# 4.13 Negotiating Committee

Leave of absence with no loss of pay and with no loss of credits shall be granted to a member of the Union who participates in negotiations, conciliation, mediation, or arbitration pursuant to Section 40 (Voluntary Arbitration) of the Labour Relations Act, provided that not more than five (5) employees at any one time shall be permitted such leave.

Members of the Union granted such leave of absence shall also be granted reasonable time off with pay and with no loss of credits to a maximum of two (2) days per member to attend Union bargaining team caucus sessions held immediately prior to negotiations for renewal of the Collective Agreement.

### 4.14 New Staff Interviews

A representative of the Union shall be given an opportunity to meet with new staff covered by this Agreement within regular hours, without loss of pay, for a maximum of thirty (30) minutes during the first three (3) months of employment in order to acquaint the new staff members with the Union and the benefits and duties of Union membership. Scheduling of such meetings will be co-ordinated through the Department of Human Resources. In the alternative and where it can be accommodated within the time frame specified above, the Union will be given thirty (30) minutes to meet with those bargaining unit staff attending scheduled New Staff Orientation Sessions.

The Department of Human Resources is to notify the President of the Union of the names of new staff members within thirty (30) days of their commencement of employment.

# 4.15 Staff Interviews

It is acknowledged that there will occasionally be circumstances where a staff member, for various reasons, has concerns about interview situations involving their supervisor. In such cases, a staff member is encouraged to bring such concerns to the attention of the OPSEU President and the Associate Vice President (Human Resources). In appropriate cases, after review and discussion of the concerns, arrangements will be made for an OPSEU representative to accompany the staff member at the interview session.

# 5.1 Discrimination and Harassment

No discrimination, interference, restrictions or coercion shall be exercised or practised by the University or the Union with respect to any staff member in regard to any matter to do with terms and conditions of employment by reason of race, creed, colour, sex, marital status, nationality, ancestry, place of origin, political or religious affiliations, sexual orientation, citizenship, age, record of offences, family status, handicap, language (unless a bona fide occupational requirement of a position), nor by reason of membership or non-membership or activity or lack of activity in the Union. The above terms are as defined in the Ontario Human Rights Code, where so specified.

Every staff member has a right to freedom from harassment in the workplace by the employer or agent of the employer or by another employee due to the prohibited grounds as stated in the Code. Harassment means engaging in a course of vexatious comment or conduct that is known, or ought reasonably to be known, as unwelcome.

A staff member is not required to perform any duties of a personal nature not connected with the approved operations of the University.

If an employee has filed a grievance under Article 5.1 and wishes to exercise rights under the university's Human Rights Policy, the grievor shall file a formal request to hold the grievance in abeyance under Article 20.6 (j).

# 5.2 <u>Employment Equity</u>

Trent University is an employment equity employer, as evidenced by its participation in the Federal Contractor's Program and the development of an employment equity policy through its Presidential Advisory Committee on Employment Equity. As a result, the Employer is committed to establishing equal opportunities of employment and a workplace environment that is free of discrimination, and it jointly undertakes with the Union, through consultation and negotiations, as well as co-operation with the sub-Committee on Employment Equity, to seek to identify and remove any barriers which may exist in employment areas, including the recruitment and hiring of, as well as professional development opportunities for, bargaining unit members.

The parties agree to work together, through the Employment Equity sub-Committee of the Presidential Advisory Committee on Human Rights, Equity & Accessibility, to consider and implement measures to improve the representation of the designated groups as defined by Employment Equity legislation and/ or by the University community and which have been found to be under-represented.

The Joint Committee shall review and approve, prior to implementation, any measures which might be recommended by the Employment Equity sub-Committee that have an impact upon rights and/or procedures set out in this Agreement.

The Union shall have the right to appoint one representative to each of the Employment Equity sub-Committee and the Presidential Advisory Committee on Human Rights, Equity & Accessibility.

# **ARTICLE 6 - ADMINISTRATION OF THE AGREEMENT**

### 6.1 <u>Relations Between the Parties</u>

If disputes arise between the parties over matters of interpretation concerning terms and conditions of employment, absence of express reference to such matters within this Agreement shall be discussed between the parties and, if resolution is not forthcoming from such discussion, differences in interpretation shall be referred to the Joint Committee and/or arbitration as appropriate.

#### 6.2 Advisors and Representatives

It is understood and agreed that either party to this Agreement may have the assistance of advisors or representatives in negotiations or other formal dealings with the other, under the provisions of the Collective Agreement or before an external Board or Commission.

### 6.3 Joint Committee on the Administration of the Agreement

Within sixty (60) days of the signing of this Agreement, parties shall establish a Joint Committee to be comprised of three (3) representatives of the Union and three (3) representatives of the University.

The Joint Committee shall function in an advisory capacity to the parties, but shall not in any way interfere with the normal functioning of any agreement between the parties, or limit the rights of the parties to enter into discussions which could result in the formation of further agreements regarding terms and conditions of employment.

The Joint Committee shall determine its own procedures, subject to the following provisions:

- 1. The Joint Committee shall be chaired jointly by a representative of the Union and a representative of the University administration, who shall together be responsible for the preparation and distribution of agenda and summary notes for meetings, and shall alternate in presiding over the meetings.
- 2. The Joint Committee shall meet at least once every two (2) months, but may meet more often by mutual agreement of the Chairs, and shall meet on five (5) days' notice at the call of either of the Chairs, or as otherwise agreed.
- 3. A quorum for the Joint Committee shall be four (4) of its members, with a minimum of two (2) members representing each of the University administration and the Union.
- 4. The parties may, by mutual consent, expand the Joint Committee or create subcommittees. The membership and procedures of

subcommittees shall be determined by the Joint Committee, subject to the condition that at least two (2) members of each subcommittee shall be members of the Joint Committee, and that the Chair of the subcommittee must be chosen from among the members of the Joint Committee serving on the subcommittee.

### 6.4 Addresses for Official Communications

Except where otherwise provided, official communication between the parties shall be addressed to:

The University

The Associate Vice President (Human Resources), Trent University, 1600 West Bank Drive, Peterborough, Ontario. K9J 7B8

# The Union

The President, OPSEU, Local 365, Trent University, Peterborough, Ontario. K9J 7B8

with copies to the OPSEU Staff Representative

OPSEU Staff Representative, 1840 Lansdowne Street W. Unit 12, Peterborough, Ontario. K9J 6X4 FAX: 741-5362 Telephone: 705-748-9100

# 6.5 <u>Term of Agreement</u>

(i) The parties herein agree that the term of the Collective Agreement shall be from the date of ratification by both parties to June 30, 2022, and that the effective date of amendments to the Agreement shall be the date of ratification, unless otherwise stipulated.

# 6.6 Renegotiation of Agreement

Either party to this Agreement may, within ninety (90) days prior to termination of this Agreement, present to the other party in writing proposed terms of a new or further Agreement and/or amendments to this Agreement, and a conference shall be held within twenty (20) days, or as otherwise agreed by the parties, at which time the parties will commence negotiations on the proposed amendments and/or terms of a new Agreement. Negotiations will be conducted in the spirit of good faith.

### 6.7 Copies of the Agreement

Within sixty (60) days following ratification of this Agreement, the University shall prepare and provide each staff member with a copy of the Agreement together with those of its appendices, which the parties agree will be distributed. The University shall simultaneously provide to the Union for its own use forty (40) copies of the Agreement. The University shall routinely provide a copy of the Agreement to newly-appointed staff members. The costs of preparation and distribution shall be borne by the University. The Union may have further copies of the Agreement beyond the forty (40) listed above, and they will be provided by the University. The Union will be billed for the University's cost of printing these additional copies.

Should the supply of prepared copies be depleted, further copies will be available electronically through the Trent University web site.

# 7.1 General Statement

The University and the Union agree that the protection of the health and safety of members and other persons in the workplace is an important matter of mutual concern and that both the University and the members have responsibilities delineated in the Occupational Health and Safety Act (OH&SA), the Workplace Safety and Insurance Act (WSIA), the Trent University Health and Safety Policy and the Joint Health and Safety Committee Terms of Reference, as amended from time to time.

The University acknowledges its obligations under the Ontario Occupational Health and Safety Act and related legislation, and in particular, its responsibility to provide a workplace where the health and safety of employees is protected as they carry out their duties. The University agrees to provide protective equipment wherever it is required for the safe and effective performance of duties, and to ensure that safety equipment, materials and protective devices are maintained in good condition. No employee shall be disciplined for refusal to perform work where the employee has acted in compliance with the Ontario Occupational Health and Safety Act. For its part, the Union agrees to promote any education programs of information and instruction initiated by the Employer and/or required by relevant legislation that will promote health and safety awareness and training among members of the bargaining unit.

As health and safety is a matter of concern to all members of the University community, any seminars or instruction to be offered will be widely advertised throughout the University and posted on bulletin boards for the information of staff.

# 7.2 Health and Safety Committee

The University acknowledges that the Union has the right to appoint three (3) representatives to the Trent University Health and Safety Committee. Staff representatives on the Safety Committee shall be entitled to a reasonable amount of time off from work with no loss of seniority or earnings to attend seminars sponsored by the government or the University or the Union for instruction and upgrading on health and safety matters. Such time off will be recommended by the Health and Safety Committee and approved by the Department of Human Resources.

The University undertakes to respond in a timely and effective fashion, in accordance with the provisions of the Occupational Health and Safety Act, to the recommendations and legitimate concerns of the Trent University Health and Safety Committee.

# 7.3 First Aid

In accordance with the Occupational Health and Safety Act and the

Workplace Safety and Insurance Act, the University shall provide for the proper training, during working hours, of at least two (2) members of each department or work location in First Aid methods, with no loss of pay. Such courses shall be offered on a regular basis, and at least annually, and shall be widely advertised to all staff. Permission to attend will not be unreasonably denied. The University agrees to encourage the broad participation of employees in such training.

First Aid Kits shall be supplied in accordance with the regulations of the above Act and their location widely advertised to staff.

# **ARTICLE 8 - STAFF INFORMATION**

#### 8.1 <u>Personnel Records</u>

For payroll and benefit plan purposes, it is imperative that the University maintain accurate personnel records. The staff member should advise the Department of Human Resources immediately if there is any change in personal data, such as marital status, number of dependents, address or telephone number.

Records of employment, including the original application form, experience and attendance histories will be kept in the Department of Human Resources confidential files. Staff members have the right to examine their personnel records by application to Department of Human Resources during normal business hours. In the event that letters of reprimand, or other materials which state or imply censure, are added to the personnel files, the staff member will be notified and provided with a copy. Such letters which are two (2) years or more old shall be removed from the file and oral reference shall not be made to them in the event of subsequent disciplinary action or future promotional opportunity, provided that no such documents relating to similar problems have been added to the file during that period.

Members may request that material which they believe to be false, unsubstantiated, scandalous or irrelevant be removed from their personnel files. Denial of such request shall be referable to the grievance procedures. Members may supplement the contents of their personnel file by placing in it statements describing and correcting any error, omission or other irregularity.

#### 8.2 <u>Confidentiality of Personnel Files</u>

Access to personnel files will be limited to staff in the Department of Human Resources and to supervisors and other authorized University officials in connection with personnel, administrative and/or labour relations matters.

Requests from external agencies for various types of information about staff should be referred to the Department of Human Resources, which will only confirm employment at Trent University and will not divulge such personal information as salary, classification, birth date, etc., without the written authorization of the staff member.

# **ARTICLE 9 - DISCIPLINE AND DISCHARGE**

#### 9.1 <u>University Responsibility</u>

The University has a responsibility to inform staff of unsatisfactory performance in a timely and open manner and also where appropriate to provide employees an opportunity to correct unsatisfactory performance. Major Infractions or termination for non-culpable circumstances may not have been preceded by previous actions.

#### 9.2 Discipline Policy

With respect to most disciplinary problems, the steps outlined in the University's discipline policy will apply.

#### 9.3 OPSEU Representation

Normally, a staff member shall have the right to be accompanied by an OPSEU representative on the occasion of an interview with a representative of management where a formal written warning, suspension or discharge is to be discussed.

#### 9.4 Notification

A staff member who receives a formal warning or suspension or is discharged shall be sent a letter confirming the reason thereof within three (3) working days of receiving oral advice. In cases of suspension, the length of the suspension or the condition for lifting the suspension will be stated. A grievance concerning a discharge may be submitted within five (5) working days of receipt of such letter. The Department of Human Resources will inform the Union of any suspensions or discharges.

# **ARTICLE 10 - EMPLOYMENT**

### 10.1 **Position Definitions**

# (a) <u>Regular:</u>

Continuing employment throughout the calendar year.

(b) <u>Recurring:</u>

Continuing employment throughout the calendar year incorporating a scheduled period of layoff of not more than four (4) months subject to normal recall rights for the position from which the incumbent was laid off.

(c) Contracted:

Employment for a definite term or task expected to last normally not more than twelve (12) months except where (a) such employment is to replace an employee on long-term leave of absence, long term disability or Workplace Safety Insurance in which cases the duration of the contract may coincide with the absence, or (b) with the agreement of the Union.

Contracted appointments are not substitutes for, or alternatives to, regular or recurring appointments. With the exception of members in the Career Development Program (Article 19) regular or recurring employees will not be considered for contract opportunities.

Contract appointments shall normally be made, for example:

- 1. while a staff member goes on leave;
- 2. while a staff member takes up an appointment involving changed responsibility;
- 3. when funds are available from sources which stipulate a contractually limited term of availability (e.g. grants), or which are not part of the department's base staffing budget;
- 4. while a staff member undertakes a training period in relation to a new position (e.g. promotion, lateral re-assignment).
- (d) Limited Term Appointments:

While recognizing that contracted appointments are not substitutes for, or alternatives to, regular or recurring appointments, with specified exceptions, contracted appointments are not to exceed twelve (12) months (as indicated in 10.1(c), above).

Notwithstanding these provisions, the University and the Union acknowledge that there are, from time to time, opportunities available

to the University to employ individuals through funding arrangements that may extend beyond twelve (12) months. Subject to the following procedures and conditions, the parties agree that in such circumstances, the University may employ up to 10 FTEs in Limited Term Appointments exceeding twelve (12) months duration at any point in time, subject to the following conditions:

- i) A limited term appointment may be made where the proportion of non-base or non-operating funding is at least 50% of the total wages/salary of the position.
- ii) Limited term appointments shall not exceed a maximum duration of sixty (60) calendar months. The termination of any employee who is an external appointment employee at the conclusion of a limited term appointment, or upon early cessation of applicable funding for such an appointment shall be understood to be a termination and shall not be deemed a violation of the Collective Agreement.
- iii) Limited term appointments shall initially be posted internally in accordance with the provisions (Article 18) of the Collective Agreement, and internal candidates (current regular or recurring employees) shall be considered prior to the consideration of external applicants.
- iv) At the conclusion of the LTA, these regular or recurring employees shall be appointed to the first available comparable position. Where an appointment is necessary to a lower level position, the employee shall be red-circled for salary purposes at the rate of pay for the position they held immediately prior to their appointment as a LTA.
- v) Where the successful candidate for a limited term appointment is currently a regular or recurring employee in the bargaining unit, and notwithstanding Article 13.14 of the Collective Agreement, any additional compensation shall be reflected in a change to base salary or hourly rate, non-statutory holidays shall be computed on the basis of the limited term rate, and seniority shall accrue on the basis of the schedule of the limited term appointment. Where a regular or recurring employee is appointed to a temporary assignment initially on a contract basis and the contract is subsequently converted to a limited term appointment, seniority shall be calculated retroactively to the original date of appointment in the contract position.
- vi) Compensation for limited term appointments shall be at the level determined by classification in accordance with the provisions of the Job Evaluation System and the provisions of the Collective Agreement, including Articles 13.7, 13.8, 13.15.

vii) Where the successful candidate for a limited term appointment is an external applicant, such employee shall be a member of the bargaining unit for the duration of the appointment and, subject to carrier approval and limitations, if any, or Pension Plan provisions, shall be eligible for benefit coverage on the same basis as regular or recurring employees. It is understood that parties will jointly recommend to the Pension Committee and the Board of Governors that the definition of Limited Term Appointments as captured in the Pension Plan (Section 2, Definitions) shall include LTAs in the OPSEU bargaining unit. Such employees shall, however, accrue seniority during the term of the appointment.

A person not subject to this Collective Agreement immediately preceding their appointment to a LTA shall have no claim to continued employment with the University by virtue of their status as a LTA.

viii) Employees engaged in limited term appointments who are not regular or recurring employees, may compete for posted vacancies. For the purposes of posted vacancies, such employees shall be considered external applicants. If successful for a posted position, the candidate shall be credited seniority and service from the commencement of the limited term appointment; the probationary period shall be waived and an assessment period substituted as referenced in Article 13.11.

Employees engaged in limited term appointments who are regular or recurring employees, may compete for posted vacancies.

- ix) Limited term appointments shall be included on information listings provided to the Union by the Department of Human Resources pursuant to Article 4.5.
- x) The current employee(s) who are in positions which will become LTAs are covered retroactively by the terms of this Agreement.

Insured benefits shall commence following ratification of this Collective Agreement, and it is understood these current employees shall not exceed 10 FTE.

Additional requests for Limited Term Positions, funded through targeted or time limited initiatives, will be brought to Joint Committee for consideration.

(e) Stand-by:

Employees working under an arrangement whereby their names are maintained on a "stand-by" list and they may work or not at their

discretion when requested to do so.

It is understood that stand-by employees will not be employed by the University for periods in excess of ten (10) consecutive working days.

Stand-by appointments are not substitutes for, or alternatives to, regular or recurring appointments.

- (f) <u>Full-time:</u> Employment for thirty-five (35) hours or more per week.
- (g) <u>Part-time:</u> Employment for less than thirty-five (35) hours per week.

### 10.2 Changes in Position Definition

If the definition of the position of an employee (Article 10.1) changes as a result of a change in the number of hours of work, , the incumbent may first exercise a right of first refusal for the modified position. Should the employee elect not to exercise a right of first refusal for the modified position the posting and relevant provisions outlined in Article 11.6 shall apply.

In the case of a multiple incumbent position, employees with more than twelve (12) months of service may, in seniority order, exercise a right of first refusal.

### 10.3 Conversions to Regular or Recurring Positions

The maximum durations of employment for a contract employee are set out in Article 10.1 and in Articles 4.12, 10.18, 13.11, 16.4 and 16.5. In the event that the employment of a contract employee inadvertently continues beyond these maximum periods, the University shall immediately terminate such employment, or if the University determines a continuing need, post and fill the position in accordance with the provisions of the Collective Agreement. Monthly listings are provided to the Union in accordance with Article 4.5.

# 10.4 Job Descriptions and Classifications

The Department of Human Resources will provide each staff member with the relevant job description, together with an outline of the classification levels and corresponding salary ranges. At the time of hiring, the staff member will be informed in writing by way of a letter of appointment, of the classification, salary range and other details corresponding to the staff member's new position.

# 10.5 Probationary Period

#### **Definition**

The initial period of employment at Trent University in a regular or recurring position during which a staff member is expected to adapt to the position and

to acquire the knowledge of techniques, procedures and skills inherent to the position. It is understood and agreed that a lesser standard of just cause for discharging employees will apply during the probationary period, including unsatisfactory performance. The University shall conduct one or more performance reviews prior to the completion of the probationary period, a copy of which shall be forwarded to Department of Human Resources, the Union and the employee.

#### **Duration**

Each new staff member is hired on a probationary period of three (3) months to six (6) months, the duration of which will be determined by the Department of Human Resources in consultation with the appropriate supervisor. The length of the probationary period will be stated clearly in the letter of appointment.

### **Notification**

Before the end of the probationary period, the supervisor will review the staff member's progress and performance. If performance is satisfactory, the appointment will be confirmed in writing by the Department of Human Resources. An initial probationary period may be extended by the employer provided the maximum does not exceed six (6) months. When a probationary employee is terminated, the Union will be notified, and the termination may become the subject of a grievance.

# 10.6 Medical Examinations

For the purposes only of assessing the ability of a candidate to perform the bona fide occupational qualifications of a job, a staff member may be required to undergo a medical examination before employment, or prior to the completion of the probationary period. Medical examinations may also be requested at other times, as required for insurance purposes. The University will assume the cost of medical examinations required for University purposes.

# 10.7 Seniority

# **Definition**

Seniority is defined as the length of continuous employment, measured in years of equivalent full-time service, with the University in any regular or recurring position, as well as contract positions fulfilled by laid-off staff (11.4). In the event a contract staff becomes either regular or recurring and the break in service is no greater than sixteen (16) weeks, their employment as a contract staff will be credited towards the employee's seniority and service, (for example vacation leave accrual and severance), once they become seniority rated.

# <u>Usage</u>

The University shall maintain a seniority list showing the date upon which each staff member's service commenced. An up-to-date seniority list shall be sent to the Union and posted on agreed-upon bulletin boards in February of each year.

Where two (2) or more staff members have equivalent full-time service, seniority will be determined by lot, as the need arises, at a meeting of the Joint Committee on the Administration of the Agreement. Seniority shall be used in situations arising from redundancy, layoffs, transfers and promotions. Seniority does not accrue during periods of layoff, unpaid leave of absence as set out at Article(s) 16.1, 16.4 and 16.5. Seniority ceases when employment at the University terminates. Layoff, subject to recall or unpaid leaves of absence, do not terminate seniority.

### 10.8 Termination

### Definition and Usage

Termination from the University is defined as the end of service of the staff member. Temporary layoff is not termination until such layoff is terminated by the staff member or through the expiration of recall rights. Staff members will not be terminated except in one of the following ways:

- (a) Retirement
- (b) Resignation
- (c) Failure to satisfactorily complete the probationary period
- (d) Termination of recall rights after layoff
- (e) Dismissal for cause
- (f) Illness or disability as defined in 10.14
- (g) Death
- (h) Completion of contracted or casual employment

#### 10.9 Retirement

In accordance with the terms of The Contributory Pension Plan for Members of OPSEU Local 365 and Exempt Administrative Staff of Trent University, a staff member's normal retirement date (NRD) will be the 1st of the month coincident with or immediately following the staff member's 65th birthday. Notwithstanding that NRD is defined in the pension plan, a staff member has the option to continue in active employment beyond age 65. Staff members who elect to continue in active employment after reaching age 65 will continue to contribute to the pension plan and will continue to accrue pension benefits until the member's chosen date of retirement. In accordance with the Income Tax Act, a member must commence receipt of retirement benefits by no later than December 1 of the year in which the member turns age 71, although the member may continue in active employment beyond that date.

The Department of Human Resources provides an estimate of pension benefits and other counselling for retirees on request. Retirees will not be re-employed by the University except in a contract, temporary or on-call capacity and with the approval of the Union. Such approval shall not be unreasonably withheld. Employees who left the University under the Early Retirement or Career Change Incentive Programs shall not be re-employed in any capacity until after their normal retirement date.

#### 10.10 Resignation

Staff members must notify their supervisor in writing not less than two (2) weeks before the effective date of resignation. A copy of such notice must be forwarded to the Department of Human Resources. Advance notice of termination is important to the University in order that it may meet its staffing needs. Notwithstanding Article 15.1 of this Agreement, staff members who do not provide notice as above will receive vacation pay on termination only to the amount required by the Employment Standards Act of Ontario.

Staff members who are absent from work for a continuous period in excess of two (2) weeks, without giving reasonable notice to their supervisor prior to the expiration of this two (2) week period, will be considered to have resigned. The University will immediately provide written notice to the staff member and a copy of same to the Union.

It is recognized that there may be exceptional circumstances which have made it impossible for the staff member to notify the supervisor about the absence. The University may agree to reinstate a staff member who has been considered to have resigned within thirty (30) days of the resignation notice, provided that within this period the staff member has submitted a request for reinstatement to the Associate Vice President (Human Resources) together with an explanation of absence (copy to the Union).

# 10.11 Failure to Satisfactorily Complete the Probationary Period

A staff member who has not satisfactorily completed the probationary period outlined in Article 10.5 may be discharged for cause.

# 10.12 Termination for Cause

Subject to express provisions elsewhere in this Agreement (Articles 10.5 and 10.8) discharge of regular or recurring staff members shall be for just cause. The Associate Vice President (Human Resources) or designate shall notify the member in writing of the effective date of discharge, the grounds for discharge and the member's right to grieve. The Union shall receive a copy of this notification.

# 10.13 Unsatisfactory Work as Just Cause for Termination

If discharge for unsatisfactory work is contemplated, written notice to the staff member allowing not less than three (3) months for the required improvement in performance will be given. The written notice, a copy of which will be placed in the employee's personnel file, will include the particulars of the work performance deficiencies and the steps required to correct such deficiencies. The University will provide reasonable job related training or equipment necessary to achieve the required improvement.

Where the required improvement has not been demonstrated resulting in a decision to discharge the staff member, the Associate Vice President (Human Resources) shall notify the staff member in writing of the pending discharge and of the member's right to meet informally and privately (with Union representation) with the Associate Vice President (Human Resources) to discuss the matter within five (5) working days of the notice. The Union shall receive a copy of this notification.

A written resignation tendered by the member within twenty-five (25) days of the first meeting to discuss discharge for unsatisfactory work will be accepted.

#### 10.14 Illness or Disability

Staff members who qualify for disability benefits under Long Term Disability or Workplace Safety Insurance shall have the right to return to their former position if it still exists or in the alternative to a comparable position if they are fit to return to work within a twenty four (24) month period from the commencement of their short term sick leave.

Beyond twenty four (24) months, when an employee is certified as able to return to work, the University will place them in a position at the same band and nature of work to the extent possible. The employee shall be given training and a reasonable opportunity to adjust to a new position.

Irrespective of the duration and the nature of the illness/disability, the University undertakes, where appropriate, to provide rehabilitative employment for which the staff member is qualified and comparable in compensation.

A staff member who is absent from work due to illness for six (6) months, but who does not qualify for benefits under the Long Term Disability plan or under Workplace Safety Insurance, shall be laid off.

An employee laid off under this clause shall elect in writing, within one week of receiving notice, either (a) to forgo recall rights and accept notice and severance as per Article 11.5 or (b) elect to be placed on unpaid leave and receive recall rights as per Article 11.7. To retain recall rights, a medical certificate must accompany any rejection of recall within ten (10) working days of the date of the registered recall letter.

#### 10.15 Death

When a staff member dies in service, the estate will receive salary payments to the end of the month following the month in which the death occurs plus vacation pay accrued up to the date of death. The Department of Human Resources will initiate action on such matters as the Trent Pension Plan and Life Insurance. Coverage in semi-private, extended health and dental plans for dependents ends the last day of the third month following the month in which the employee dies.

### 10.16 Completion of Contracted Employment

Contract employees are entitled to paid holidays in accordance with the provisions of Article 15.3; and to 4% vacation pay or one (1) day per month to a maximum of ten (10) days over twelve (12) months.

The employment of contract employees can be terminated at any time on the giving of one (1) week's notice or pay in lieu of notice if the employee has been employed continuously for more than three (3) and less than twelve (12) months; or two (2) week's notice or pay in lieu if employed continuously for more than twelve (12) months.

### 10.17 Severance Pay on Termination

Resignation, Dismissal, Retirement, and End of Contract Employment.

Staff members who are terminated for reasons of resignation, dismissal, retirement, or end of contract employment shall not be the recipients of severance pay.

### 10.18 Technological Change

#### **Definition**

In the event that the University decides to introduce new equipment or material, or to implement a change in the manner in which it carries on its operations that is directly related to the introduction of that equipment or material, and where such introduction or implementation would be likely to adversely alter a position classification or displace any staff member, the University shall give the Union three (3) months' notice. The parties shall consult as soon as possible to review the impact on staff members and possible measures to reduce the adverse effects of the pending change.

The notice mentioned above shall be given in writing and shall contain pertinent data including:

- (a) the nature of the change;
- (b) the date on which the University proposes to effect the change;
- (c) the number, type and location of staff members likely to be affected by the change;
- (d) the effects the change may be expected to have on the staff members' working conditions and terms of employment; and,
- (e) other pertinent data relating to the anticipated and foreseeable effects and repercussions on employees.

Prior to any new regular or recurring staff members being hired to operate the

new equipment or to deal with the new technology, the University will, where necessary, first allow existing staff members a training/assessment period up to six (6) months to acquire and demonstrate the knowledge, skill and/or qualifications necessary to cope with the change, provided they are qualified by education, aptitude and/or experience. Staff members to be retrained will be selected on the basis of seniority in the following order of priority:

- 1. those who are adversely affected as outlined above;
- 2. those most suitably qualified staff members from elsewhere in the University;
- 3. those most suitably qualified staff members from laid-off staff.

Contract and/or temporary employees may be engaged for the duration of any such training/assessment period.

### 10.19 Compensation During Retraining

Staff members to be retrained will not suffer a reduction of salary/hourly rate during the training period.

#### 10.20 Salary Compensation for New Position

Pay levels for the new position following completion of the retraining will be determined through job evaluation. If the classification of the affected job is altered as a result of the changes in 10.18, the appropriate pay level shall be established following completion of the training/assessment period.

#### 10.21 Alternate Employment

If the measures decided in Article 10.18 do not reduce or delete the adverse effects and no alternate employment is available, the affected staff member(s) may be laid off pursuant to the provisions of the Collective Agreement.

# ARTICLE 11 - REDUNDANCY AND LAYOFF

# 11.1 <u>Contracting Out</u>

A bargaining unit employee cannot be reduced in hours, laid off, and/or declared redundant in order to contract work out that is currently performed by members of the bargaining unit if the primary duties of that position are ongoing and required by the University.

#### 11.2 Contracting In

Whereas there is a potential for some services and functions, currently contracted out or potential future contracts, to be performed by bargaining unit employees, the Parties agree to the following undertakings that will be carried out within six (6) months of ratification:

- 1. The University shall provide the Union with a list of services and/or functions provided by contractors outside of the University and an outline of known potential future work.
- 2. For those services which the Union wishes to explore and which would reasonably be considered to be potential work of the bargaining unit, the University will provide documentation, including any Requests for Proposals on file relative to the work being performed as long as there is no issue of confidentiality of the information.
- 3. The Parties agree to assess such services and functions on a business basis reflecting the financial, administrative, service and staffing implications.
- 4. The Joint Committee on the Administration of the Collective Agreement (Article 6.3) will conduct the assessment(s).
- 5. If after full and complete consultation by the Parties, the University determines that the work will continue to be performed by outside contractors, the University shall convey to the Union the reasons for its decision.

Future issues, not anticipated during the above review, may be raised by either Party at Joint Committee meetings.

#### 11.3 New Work

When new initiatives on campus or any work location in Ontario of the University have the potential to result in ongoing employment opportunities, the Parties will review the nature of the work for possible inclusion in the bargaining unit, if such work has not already been included. The Joint Committee on the Administration of the Collective Agreement (Article 6.3) will conduct this review. In the event the Parties are unable to agree on the application, implementation or administration of Article 11.3, the Parties agree to refer the dispute to mediation/arbitration for resolution.

#### 11.4 Layoff

It is agreed that there will be no permanent layoffs, save and except as set out below.

Prior to effecting layoffs of any regular or recurring employee, the University will take the following steps in sequential order as outlined in Article(s) 11.4, 11.5, 11.6 and 11.7.

- 1. Within two (2) weeks of the determination that there is a need to reduce the workforce, a meeting will be arranged between the Parties to review the circumstances and to mutually explore options.
- 2. The Parties will explore known, upcoming attrition such as retirements, resignations, end of contracts and existing vacancies to determine if there is an employee who is leaving or has left the work place whose work could be done by a qualified employee who otherwise will be laid off. Such employee will be placed in the position.
- 3. Next, the Parties will examine the list of employees holding contract positions, or external applicants to Limited Term Appointments (LTA) or who are on probation to determine whether the redundancy can be resolved by terminating one or more such positions and redeploying qualified regular and/or recurring employees into the positions.
- 4. The Parties will explore the possibility of a short-term layoff of no more than twenty-six (26) weeks. Should this meet the needs of the University, employees will be laid off and provided a Supplemental Employment Benefit to top up Employment Insurance Benefits to 80% of normal wages.
- 5. If after having explored the above steps, it is determined by the University that a layoff will be required, the least senior employee in the redundant position will be identified and provided with notice of layoff.
- 6. An employee who has more seniority than the identified employee may, within one (1) week of notice under step 5, volunteer to take the layoff if the junior employee has the minimum required qualifications (as defined in Article 18.2) to perform the senior employee's work. In such case, the exchange will take place as soon as possible, and the senior employee will be laid off and will receive the remainder of notice of layoff in lieu, plus their severance entitlement, but shall have no recall rights. This provision does not apply if the senior employee has unique skills which cannot be replaced in up to three (3) months.

### 11.5 Notice and Severance

Notice of layoff, inclusive of statutory rights, are as follows:

- a. an employee with less than 5 years of seniority 3 months
- b. an employee with 5 years or more of seniority 6 months

In addition to statutory rights, an employee is eligible for Enhanced Severance pay as follows:

- one week per year of service to a maximum 13 weeks

The Parties may mutually agree to alternate notice and severance arrangements or other short-term plans.

Notice shall be in writing and the notice period commences from the date on which an employee receives their written notice. The Union will be copied on the notice.

Pay in lieu of notice or severance payments end at the earlier of age 65 or date of unreduced early retirement.

### 11.6 Redeployment/Displacement

An employee who has been given notice of layoff will elect in writing, within 1 week of notice, either (a) (b) or (c) as follows:

- (a) accept the layoff and receive notice and severance payments. A lump sum payment is made to the employee in the amount of the notice and severance entitlement and the staff member forgoes their recall rights and the employment relationship is deemed to have been terminated.
- (b) to be given priority consideration to be placed in a position which comes open, subject to having the required qualifications. If no such vacancy exists, at the employees' current pay band, the employee can exercise bumping rights to displace a regular or recurring employee with less seniority in an equivalent or lesser banded position in which the displaced employee meets the required qualifications (as defined below). Employees who are redeployed into a position will receive the rate of pay of the new position at the end of the notice period. The displaced employee will have bumping rights for the duration of the working notice period. If the employee is unsuccessful in being placed in a position, at the end of the notice period, the employee will be laid off and will be given recall rights as per Article 11.7.
- (c) if there is a potential future position at the University for which the University deems an employee eligible, with training, the employee may elect to enter into a career counselling and training opportunity under the provisions of Article 19. The employee may request time away from normal work, subject to reasonable operational needs as

defined by the manager, to attend a course. Such employee may apply or be redeployed to related job opportunities which arise. If at the end of his/her notice period he/she has not been successful in obtaining a position or been placed into a position, then he/she is moved to the recall list. The training time is deemed notice under the terms of Article 11.5 and the *Employment Standards Act*.

A regular or recurring employee who is displaced by an employee under Article 11.6(b) will also have the options as laid out in Article 11.6.

For the purposes of Article 11, Job Security, the required qualifications (as defined in Article 18.2) is expanded to mean that the employee has either the education established for the position or the equivalency achieved through education and experience. The employee must be able to perform the core responsibilities of the job at an entry level. Job testing and interview(s) may be used to assist in determining an employee's qualifications. The employee must meet any requirement established by law or regulation.

#### 11.7 Recall Rights

Staff members shall have recall rights for eighteen (18) months. He/she will be offered a position which comes open and will be recalled in order of their seniority, provided they have the required qualifications (as defined in Article 18.2).

Notice of recall shall be made by registered mail to the last known address of the staff member, alternatively members may elect in writing within one (1) week of being placed on recall, to receive notice of recall by email to either their Trent or Personal email. A copy shall be sent to the Union. It shall be the responsibility of the laid-off staff member to keep the Department of Human Resources informed of a current address.

If recall takes place prior to the expiration of the period of notice of layoff or pay in lieu thereof or severance pay, all payments will be discontinued from the date of recall.

If a staff member has been laid off for a period of eighteen (18) months, without having been recalled, the employee will be provided severance payments, and the employment relationship of that staff member shall be deemed to have been terminated.

If a laid-off staff member fails to notify the University of an intention to accept recall within ten (10) working days of the date of the registered recall letter, the staff member will be considered to have resigned and the employment relationship of that staff member will be deemed to have been terminated. See Article 10.14 re: medical certificate requirements for ill or disabled members on layoff refusing recall.

If the staff member, having notified the University of an intention to return to work as provided in this Article, fails to return to work within three (3) days of

the scheduled return, the staff member will be considered to have resigned and the employment relationship of that staff member will be deemed to have been terminated.

A laid-off staff member with recall rights who accepts a definite term temporary assignment and who has been given written notice of such term at the time of appointment, is entitled to two (2) weeks' notification or pay in lieu thereof if the assignment is terminated prior to the completion date of said definite term temporary assignment as stated in the written notice. In the event a laid-off staff member with recall rights accepts a temporary assignment, his/her recall rights remain in effect in accordance with the provisions of this Article.

### 11.8 Emergency Temporary Layoffs

The Parties acknowledge that, in the event of an unforeseeable disaster beyond the control of the University, it may be impossible for a staff member(s) to work. In such cases, the University is not obliged to honour notice provisions as referenced in 11.5.

Emergency temporary layoffs shall not exceed six (6) consecutive weeks.

#### 11.9 Labour Disputes

Where the University's normal operations are disrupted due to a work stoppage resulting from a labour dispute involving members of another bargaining unit, layoffs may take place after two (2) weeks' notice or pay in lieu thereof, and members so laid off will be entitled, subject to approval by Employment and Immigration Canada, to participate in an EI Sub Plan.

# 12.1 Standard Work Week

Normally the standard work week for full-time staff consists of seven (7) hours per day, thirty-five (35) hours per week. Standard working hours are those working hours existing in any University department prior to the election of modified work week arrangements as an alternative. Individual working schedules for full-time and part-time staff are determined by supervisors and are subject to departmental needs and approval. The hours of work shall be stated in the job posting and will be defined in the appointment letter.

If a staff member has special and legitimate reasons which would justify some variation in the standard work day/work week, these reasons shall be discussed with the supervisor and approved by the Associate Vice President (Human Resources). Reasonable efforts shall be made to accommodate such requests.

In certain departments, hours of work will vary for special workload reasons. Alterations in schedules or work will be posted whenever possible in the staff member's department or office. Compensation for overtime is in accordance with Article 13.3.

Except in respect of variations in the standard work day/work week above, where a modified work week is instituted pursuant to Article 12.2, or where normal operating hours include Saturday and/or Sunday, the University normally shall schedule for each staff member at least two (2) consecutive days of rest per week.

# 12.2 Modified Work Week

A modified work week may include: flextime, banked time, or a compressed work week. The terms and conditions thereof shall be negotiated with the Union and reflected in a Letter of Understanding.

# 12.3 Overtime

Overtime must be approved in advance by the supervisor and recorded within the department. An employee will qualify for overtime after working over seven (7) hours on a scheduled work day or eight (8) hours on a scheduled work day or thirty-five (35) hours or forty (40) hours depending on the normal full-time work week for that job classification. Compensation for overtime is stated in Article 13.3.

Overtime should be necessary only in situations of limited duration. Every effort should be made to avoid overtime on a continuing or scheduled basis. The University will attempt to allocate overtime work on an equitable basis among readily available qualified staff members who normally perform those duties. In certain departments, a regular day may be adopted in excess of eight (8) hours, but not in excess of twelve (12) hours. (The total of regular plus overtime hours scheduled normally for a staff member shall not exceed twelve (12) in a day or forty-eight (48) in a week).

In cases of emergencies only, the University may require the maximum limit of hours in the day to be worked to be exceeded only so far as to avoid serious interference with the ordinary working of the University.

An employee who works three (3) or more hours of overtime immediately before or immediately following normal hours of work shall be provided with a break of not less than one-half (1/2) hour for eating purposes. Eating time will be included in the compensation calculation of overtime hours worked.

#### 12.4 Lunch Period

A break of not less than one-half (1/2) hour per day or shift will be granted to staff for eating purposes. Such time shall not be included in the above calculation of hours worked. Scheduling is at the discretion of the supervisor subject to the needs of the department.

### 12.5 Rest Periods

Two (2) fifteen (15) minute rest periods will normally be provided, one preceding and one following the eating period. Scheduling is subject to arrangements with the supervisor. Rest periods worked shall not be calculated as overtime worked nor can they be banked to accumulate paid time off.

#### 12.6 Scheduled On-Call

Staff designated as "on call" will carry a communicator and/or remain in proximity of a telephone. Such staff shall be paid at the rate of one hour per day for being "on call" whether or not they are called.

Effective January 1, 2020, and subject to the Employment Standards Act requiring 3 hours of "On Call Pay", staff designated as "on call" will carry a communicator and/or remain in proximity of a telephone. Such staff shall be paid at the rate of 3 hours per day for being "on call" whether or not they are called.

# 12.7 Working During Emergency Closure

If the University enacts an Emergency Closure, as defined in the University Class Cancellation and Closing Procedure, employees shall be paid for those days or partial days the University is closed, provided the closure occurs on a normal working day on which the employee is scheduled to work. Those employees who are required to work, as directed by their Supervisor, during a period where the University has closed operations are entitled to equivalent time off at a later day, but not to overtime pay.

The authority and responsibility for cancelling classes and/or closing the University rests with the Vice-President, Finance and Administration in Peterborough and/or the Head, Trent University Durham. Emergency closings will be firm and not left to the discretion of individuals or departments. Normally, only those staff identified in the Trent University Emergency Management Plan will be allowed on campus.

This provision does not apply to Class or Exam Cancellation.

# **ARTICLE 13 - COMPENSATION**

## 13.1 Pay Equity

OPSEU and the University acknowledge that the Job Evaluation System, implemented February 2001, and resulting Salary Grid have been built upon principles of pay equity.

### 13.2 Salary Grid and Salary Bands \*

[\*See Schedule A]

- .01 The Salary Grid will consist of thirteen (13) bands, each with seven (7) steps, and each band will span a range of thirty-one (31) points starting at one hundred (100) points.
- .02 The points assigned to each position will determine which salary band applies to the position.

"Red-circled Salaries" are those which exceed Step 7 within the correct band. Until such time as the "Red-circled" salary is exceeded by Step 7 within the correct band, employees whose salaries are "Red-circled" shall not be eligible for any scale increases which may be negotiated from time to time. When the "Red-circled" salary falls below, or becomes equal to Step 7, it will be immediately converted to a Step 7 salary, and at such time the "Red-circle" designation will cease.

# 13.3 Overtime

Overtime is defined in Article 12.3. Approved overtime shall be compensated either by pay at one and one-half (1-1/2) times the staff member's regular rate, or by time off at a rate of one and one-half (1-1/2) hours for each overtime hour worked, the method to be mutually agreed upon by the supervisor and the employee.

Where it is agreed that overtime will be paid out, employees shall submit timesheets for pre-approved overtime accrued during the previous pay period to their supervisor by no later than one (1) week prior to the published payroll deadline. Employees who submit within this timeline shall receive overtime pay in the next pay. Approved overtime to be paid out shall not be banked for a period greater than three (3) months. Time off for overtime worked cannot be banked beyond a three (3) month period without the prior written approval of the Department/College Head at the time overtime is worked.

In the event a Department offers overtime on a time-off only basis, acceptance of such overtime shall be voluntary.

# 13.4 Emergency Call-in

Emergency call-in will result in compensation to the greater of four (4) hours at regular compensation or overtime rates for the call-in hours worked, except where any of those hours in either case overlaps and extends into the scheduled working hours.

Effective January 1, 2020 and subject to the provisions of Article 12.6 changing on the same date, for call ins 3 hours or less in duration, no additional compensation beyond the "On Call" pay will be provided. Calls greater than 3 hours in duration, will result in compensation to the greater of four (4) hours at regular compensation or overtime rates (whichever applies) for the hours worked, except where any of those hours in either case overlaps and extends into the scheduled working hours.

# 13.5 <u>Supplementary Payment</u>

- .01 In addition to the basic salary to which an employee is entitled pursuant to the Salary Grid, employees may be awarded supplementary payment of one or more of the following types:
  - (a) Shift Differential Bonus
  - (b) Market Differential Bonus
  - (c) Temporary Assignment Bonus (see 13.14)
- .02 Shift Differential Bonus

Employees whose regularly scheduled hours of work are at least fifty (50) per cent outside the hours of 08:00 hours and 16:00 hours will receive a Shift Differential bonus of sixty-five cents (65¢) per hour for all the hours of the shift.

.03 Market Differential Bonus

Incumbents who are currently receiving Market Differential Bonuses shall continue to receive such bonuses, provided they continue in the position to which said bonus applies. The bonus shall be adjusted by the same percentage as any general wage increase which may be negotiated from time to time. Establishment of any new Market Differential Bonus or the application of same to any new employee, shall be at the sole discretion of the University.

#### 13.6 Salary Reductions

No salary shall be reduced as a result of promotion or lateral reassignment.

# 13.7 Ongoing Administration of the Salary Grid

.01 All new employees will normally be hired at Step 1, and all employees will, subject to satisfactory performance, advance through the Salary Grid by way of annual increments until Step 7 is attained. The first annual increment for a new employee will be awarded on July 1st, provided the employee was hired prior to the previous September 30th. For employees hired between October 1st and the following June 30th,

the first annual increment will be awarded on January 1st following the June 30th. Subsequent annual step increases will be awarded each July 1st.

New employees may be hired above Step 1 taking into account pertinent previous experience. Any such rate must be a rate in the band established for the position. The Department of Human Resources shall notify the Union of any such exception(s) through a copy of the new employee's appointment letter.

When a new employee is hired into a multiple-incumbent position, they shall be placed on the salary grid at least one step lower than any other employee in the position, except with the mutual agreement of the parties.

#### 13.8 Step Increases

- .01 The parties hereto acknowledge that regular annual step increases will be awarded as specified in this document during the term of the Collective Agreement expiring June 30, 2015, and eligible employees will ordinarily receive such increases in the normal course.
- .02 The awarding of a step increase is, however, conditional upon satisfactory performance.

It is understood that serious employment deficiencies such as may be identified in relation to disciplinary action under Article 9, will normally result in the withholding of a step increase, and could ultimately lead to Dismissal for Cause proceedings under Articles 10.12 and/or 10.13.

- .03 Decisions to withhold a step increase may be the subject of a grievance.
- .04 OPSEU will be advised in advance, and in writing, of all cases where it is decided that a step increase is to be withheld.

#### 13.9 Annual/Hourly Rates

.01 The annual rates shown on the Appended Salary Grids are based on a 35-hour work-week. The annual salaries for employees on other than 35-hour work-weeks will be appropriately computed based on the hourly rates set out in the Appendices.

#### 13.10 Promotions

Promotion is defined as the movement of an employee to a position on a salary band higher than the band associated with his/her previous position. It can result either from a re-assignment by the University, with the approval of the Joint Committee, to a position on a higher band, or from the employee succeeding in a competition for a position on a higher band.

Movement to a position with a higher point rating within the same band shall not be regarded as a promotion, but as a lateral re-assignment.

Employees who are promoted shall be assigned the salary on the new band which corresponds to at least one step greater than the salary on their previous band, except where the previous salary is below Step 1 of the new band. In the latter case, the employee shall be assigned the Step 1 salary. For employees whose salaries are red-circled, they shall, upon being promoted, be assigned the corresponding salary (if there is one) on the new band. If there is no corresponding salary, they shall be assigned to the closest salary step on the new band above their previous salary. If the red-circled salary is above Step 7 of the new band, at the time of the promotion, it shall remain a red-circled salary at the previous rate.

### 13.11 Assessment and Reassignment

Staff members who are successful candidates in a promotion/lateral re-assignment shall be considered to be in an assessment period for three (3) to six (6) months, depending on the time stated in the appointment letter. The assessment period commences when the employee is fully released from his/her former position and the obligations of 18.4. As part of the assessment period, appropriate orientation and job instruction will be provided by the University. The purpose of the assessment period is to provide the employee with a reasonable period in the new position to demonstrate that he/she can perform the job to the employer's satisfaction. Unsuccessful completion of the assessment period will result in the following:

- (a) The employee will be re-assigned to his/her former position if the position is available; or if the former position is not available, to another comparable position, if such a position is available, for which the employee possesses the minimum required qualifications including education, experience, and the ability to perform the duties of the job; or, if such comparable position is not available, to any position which is available for which the employee possesses the minimum required qualifications including education, experience, and the ability to perform the duties of the job.
- (b) The salary/wage rate to be used at the time of re-assignment will be based on the actual salary/wage earned prior to the assessment period in the new position. An employee reassigned to a lower classification than that of his/her position prior to the assessment period shall have his/her salary "redcircled" at the salary/wage earned prior to the assessment period.
- (c) Employees who are successful candidates in a promotion/lateral transfer may elect within a period of two (2) calendar months

from the employee's starting date in the new position to be reassigned as per 1(a) of this Article, and if the employee does so, he/she shall be precluded from applying to fill any other vacancy for a period of twelve (12) calendar months.

(d) It is understood that where an employee is to be re-assigned as per this Article, that where the employee's former position is filled by a contract or probationary appointment, such appointment may be terminated immediately and without notice to provide a position for the re-assigned employee, notwithstanding any other provisions of this Agreement.

# 13.12 Demotions

A demotion is a re-assignment by the University to a lower-rated classification and may occur as a result of unsatisfactory performance or other job-related reasons considered appropriate by the University. Wages shall be appropriately adjusted. Demotions may be the subject of a grievance.

### 13.13 Assignments Outside the Bargaining Unit

With the agreement of the parties, an employee accepting any assignment outside the bargaining unit shall, if returned to the bargaining unit within twelve (12) months either by action of the University or by exercising his/her right to return for any reason within twelve (12) months, be permitted upon reentry to count the service outside the bargaining unit for seniority purposes. It is further understood that for a period not to exceed twelve (12) months, such an employee shall retain the right to return to the position occupied at the time of the assignment.

# 13.14 Temporary Assignments

- .01 When a staff member is absent and another staff member of lower classification is to be assigned a significant portion of the responsibilities of the absent staff member, vacated as a result of absence for a period in excess of ten (10) working days, then such other staff member shall receive additional compensation during the period of performing those duties. Such temporary upgrades, detailing the start date and any reclassification of pay rates, must be agreed in writing with the department manager prior to the staff member assuming such duties.
- .02 Employees may be assigned a special project for a limited time that is at a higher band than their regular duties. The details of this assignment must be agreed to in writing with the department manager prior to the staff member assuming such duties. The duties will be submitted to, and evaluated by, Human Resources to determine the rate of pay. The Union will be notified by Human Resources within 10 working days of the commencement of the new assignment. Such assignments shall not exceed 12 months in duration.

- .03 Additional compensation will be based upon the rate which is at least two (2) steps higher than the employee's current salary. Further, in cases where the absent employee is more than one band above that of the employee on temporary assignment, the Associate Vice President (Human Resources) shall review the duties being performed and may, in appropriate cases and depending upon the extent of the duties being performed, award additional compensation beyond the level specified above.
- .04 Payment shall commence effective from the date on which the staff member performs such duties described in .01 or .02 above and continues for the duration of the time so performed by said staff member.
- .05 Such additional compensation will be added to the staff member's compensation as a gross pay adjustment rather than as a change in base salary or hourly rate. Such an addition will preclude additional contributions being made to pension and insurance plans. Non-statutory holidays (Article 15.3) to which an employee is entitled shall be paid to an employee on temporary assignment at the base rate of pay applicable to the employee's regular appointment.
- .06 If the staff member should not return to work in the appropriate time period outlined in this Agreement, then the job will be posted in the normal manner and applications invited from all interested staff.
- .07 If the temporarily upgraded staff member is the successful applicant, the staff member enters the new job at the greater of the appropriate start rate for the staff member in the new job or the temporary rate established in .02 above and receives salary and performance reviews at normal intervals.

#### 13.15 Performance Appraisal

Performance appraisal is the process by which the job performance and effectiveness of a staff member is determined and communicated to the member. It is the responsibility of supervisors to complete and sign assessments of the effectiveness of the members' performance of their assigned responsibilities. Performance appraisal shall be made at the completion of the probationary period/assessment period, and at least annually thereafter in conjunction with the salary review.

Performance appraisal proceedings will be initiated when the Department of Human Resources forwards individual performance appraisal forms to the Department/College Heads. It shall be the responsibility of the Department/College Head to invite the appropriate supervisor to complete the appraisal form. An interview shall take place between the supervisor and/or Department/College Head and the staff member as part of the performance appraisal review. The staff member will be encouraged to complete the section of the form requesting the staff member's comments. The staff member will sign the appraisal form acknowledging that the staff member has seen the contents in the appraisal form. Each staff member shall be given a copy of his/her performance appraisal after completion.

The Department/College Head shall review the appraisals and shall forward the performance appraisal forms to the Department of Human Resources for administrative purposes and for the maintenance of members' personnel files. All comments from supervisors and staff members shall be confidential unless and until such time as they are used for personnel decisions.

#### 13.16 Payroll Matters

#### Pay Periods and Pay Days

Salaries for staff are automatically deposited to the staff member's bank account on the twenty-eighth (28th) day of each month. If the twenty-eighth (28th) day falls on a weekend or a Statutory Holiday, salaries are deposited in advance.

Wages for hourly-paid staff are automatically deposited to the staff member's bank account on alternate Fridays, and include payment for work performed up to and including the previous Saturday.

Each staff member receives a pay statement listing deductions. Questions regarding payroll deductions should be referred to the Payroll Office, Department of Human Resources, Blackburn Hall.

The University reserves the right to tender the contract for provision of payroll services, change the supplier, etc., provided only that prior notification is provided to the Union and a comparable level of payroll service is maintained.

#### Vacation Pay for Recurring Employees

Recurring employees (with a regularly-scheduled period of layoff of not more than four (4) months, subject to recall to the position from which he/she was laid off) shall have the following options with respect to vacation pay:

- (a) to receive their vacation pay at the time of layoff; or
- (b) to request, in writing, to delay receipt of the amount owed at the time of layoff and receive such vacation pay within two (2) pay periods of their return to work.
- (c) to request in writing to have a vacation day with pay subject to Article 15.1.

The University is not accountable for any issues of eligibility for external programs arising from a delay in vacation payment.

# 13.17 Payroll Deductions

#### **Statutory Deductions**

Certain deductions are required by law. Other deductions are required to fund benefit coverage. Some deductions (e.g. Canada Savings Bonds, charitable, etc.) require prior approval by staff members. New staff members are advised of the deduction amounts during their benefit orientation interview.

# 13.18 Garnishees, Assignments, Third-Party Demands

Legal requirements authorizing deductions from a person's pay for reasons of a court garnishee or government third-party demand must be honoured by the University.

# 14.1 Benefits

As a minimum, and subject to the eligibility and other detailed provisions of the various group benefit plans (both statutory and non-statutory) which are summarized below, staff members shall be provided with the benefits listed. In the case of statutory benefits (EI, CPP, WSIB), further information is available through the various government offices and/or through the Department of Human Resources. For non-statutory insured group benefits, staff members should consult their benefit booklet. Further information is also available through the Department of Human Resources.

# 14.2 Statutory Benefits

- (a) <u>Employment Insurance (EI)</u> Board/employee premiums as determined by the Federal Government
- (b) <u>Canada Pension Plan (CPP)</u> Board/employee premiums as determined by the Federal Government
- (c) <u>Workplace Safety and Insurance Board (WSIB)</u> Board premiums as determined by the Province

# 14.3 Insured Group Benefits

- (a) <u>Semi-private Hospital Coverage</u> Board pays 100% of this coverage
- (b) <u>Extended Health Care</u> Single or family coverage, as appropriate; \$25 single/\$50 family deductible, paid by staff member; Board pays 100% of premium; coverage includes, but is not limited to, the following:
  - (i) Prescription drugs
  - (ii) Private nursing where ordered by attending physician
  - (iii) Difference in cost between semi-private and private (where available) hospital accommodation
  - (iv) Contact the Department of Human Resources regarding details of further coverage which is provided, including ambulance, prosthetic appliances, speech therapy, etc.
- (c) <u>Additional Extended Health Benefits</u> Board pays 100% of the premium costs for the following benefits:
  - (i) Vision Care a four hundred and fifty dollar (\$450.00) optical benefit which shall be available once in any twenty-four (24) consecutive month time period in accordance with the terms and conditions of the Plan.

Eye Examinations – a fifty dollar (\$50.00) eye examination benefit which shall be available once in every twenty-four (24) consecutive month time period in accordance with the terms and conditions of the Plan.

- (ii) Hearing Aids up to \$750 every four years per family member
- (iii) Massage Therapy up to \$750 per year per family member
- (iv) Chiropractic Coverage up to \$500 per year
- (v) Physiotherapy Coverage- up to \$1250 per year
- (vi) Naturopathic Care up to a maximum of \$300 per year
- (vii) Psychological Care- up to a maximum of \$700 per year
- (viii) Out-of-Country Coverage effective September 1, 2004

### (d) Dental Plan

- Dental Plan single or family coverage, as appropriate, based on the previous year's Ontario Dental Association (ODA) Schedule, updated annually each January 1
- (ii) Dental Plan/Orthodontic Treatment single or family coverage, as appropriate; \$2,000 lifetime maximum per eligible family member; 50% co-insurance (i.e., carrier and staff member pay half each of eligible orthodontics fees)
- (e) <u>Group Life Insurance</u> term insurance equal to 1.5 x annual salary (Board pays 100%) - plus optional coverage of an additional 1x, 2x, 3x, or 4x annual salary, (Employee pays 100% of premium on options.)
- (f) Long Term Disability Board self-insures for absence from sickness/injury for six (6) months; after six (6) months, where total disability, LTD coverage provides for 80% of monthly insured earnings; these LTD benefits are taxable income.

# 14.4 Trent Pension Plan

It is understood and agreed by the parties that the Contributory Pension Plan for OPSEU and Exempt Employees of Trent University" (Amended July 1, 2001) as amended from time to time, forms part of this Collective Agreement and, insofar as it relates to members of the OPSEU bargaining unit, may only be altered or amended by mutual agreement of both parties. Copies of the Plan text, as well as additional information about its provisions, are available to any member of the Plan, upon request, through the Department of Human Resources. By way of summary, staff members who participate will contribute 9% of pensionable earnings until June 30, 2020 at 11:58pm, effective June 30, 2020 at 11:59pm staff members will contribute 9.8% of pensionable earnings. The University contributes required amounts in accordance with Appendix XIII (Items 5 and 6). Pension benefits, which are entirely separate from Canada Pension Plan entitlements, provide for an annual pension commencing at Normal Retirement Date (the first day of the month coincident with or next following the date the member attains age 65) equal to 2% of final average earnings times the number of years of service. Effective July 1, 2013, final average earnings are determined by calculating the average of the best consecutive five (5) of the last ten (10) years of the member's earnings at the University. In no event, should the average earnings be less than the average of the highest three consecutive years out of the last ten years as of June 30, 2013.

Additional pensionable earnings shall include hours worked by part-time or recurring employees beyond the base hours of their regular position to a maximum of the full-time equivalent. It does not include additional contract positions.

### 14.5 Tendering of Benefits Contract

For those insured benefit plans which are non-statutory, the Board reserves the right to tender the benefits contract, change the carrier, etc., provided only that equivalent benefit levels are maintained.

#### 14.6 Coverage

A summary of benefit participation by position classification is shown in Article 14.8.

The following information relates to benefits coverage on Termination of Employment:

(a) Employment Insurance (EI)

Staff members will be provided with a Record of Employment upon termination of employment or if there is to be an interruption of their earnings of seven (7) days or more. The information on the Record of Employment is used by Employment and Immigration Canada to decide if there is an entitlement to benefits and the rate and duration of benefits.

(b) <u>Semi-Private Coverage</u>

Coverage in this plan ceases at the end of the month following the month in which a staff member terminates.

(c) Dental Plan

Coverage in this plan ceases at the end of the month following the month in which a staff member terminates.

(d) Extended Health Care

Coverage in this plan ceases at the end of the month following the month in which a staff member terminates.

(e) Long Term Disability Insurance

Coverage ceases immediately with termination of employment. Benefits payable by the insurance company cease when employment terminates due to the death of staff members, or if staff members reach normal retirement age, or if they cease to be totally disabled, except that for the two (2) year period immediately following the six (6) months during which the University maintains salary payments according to sick leave policies, staff members may engage in "rehabilitative employment" and continue to receive partial benefits. The Department of Human Resources can be contacted for further information on benefits as outlined in the insurance policy.

#### (f) Group Life Insurance

Coverage ceases upon termination of employment. However, application for private insurance coverage may be made without providing evidence of insurability provided application is made to the group life insurance company within thirty-one (31) days of date of termination. The Department of Human Resources can be contacted for further details.

(g) Trent Pension Plan

Members of the Plan who terminate employment with the University are entitled to termination benefits as outlined in the Pension Plan text.

#### (h) Canada Pension Plan (CPP)

The Canada Pension Plan, to which all staff members between the ages of 18 and 65 must belong from their first day of work, provides for payment of a pension, based on the amount of individual earnings (up to a certain maximum), and service from January 1, 1966. Death and survivor benefits are also payable from this plan, and under particular circumstances, a disability pension may also be available.

At age 65, staff members may make application to receive a Canada Pension benefit (in which case contributions will cease) or they can postpone applying for a pension, and, provided they are working, continue to pay contributions up to age 70. At age 70, their contributions will cease and they can apply for their pensions in full, whether or not they keep on working and regardless of their earnings. It is possible to apply for Canada Pension benefits any time after age 60 and before age 65, but there is a 1/2 per cent per month deduction for each month of early retirement prior to age 65.

#### (i) <u>Workplace Safety Insurance</u>

The University pays premiums annually to the Workplace Safety and Insurance Board (WSIB) based on an amount per \$100.00 of assessable earnings for each employee in accordance with the rates set annually by the Workplace Safety and Insurance Board.

The Workplace Safety and Insurance Board provides compensation and/or medical aid to employees who, as a result of an accident related to their work, are injured to the extent that they experience a loss of time from work.

Employees who suffer an injury at work must immediately report this injury to their supervisors who in turn must immediately report this injury to the Health and Safety Office (Department of Risk Management). This procedure should be followed whether or not the injury is severe enough to warrant loss of time from work and/or medical aid. A report is filed with WSIB immediately. Such a procedure aids in the establishment of claims at a later date, if necessary, and in the correction of safety hazards in the University work place. The University is required to pay fines to WSIB for late filing of accident reports.

# 14.7 Coverage and Cost Sharing

#### <u>Coverage</u>

Legislation and the applicable plan policies in effect at the date of signing of this Agreement define the terms of participation and level of benefits provided under Articles 14.1, 14.2, 14.3, 14.4 and 14.6 of this Agreement. A summary of participation by position classification is shown as Article 14.8.

#### Cost Sharing

The University and employees will contribute as specified to the cost of providing the benefits as outlined in Articles 14.1, 14.2, 14.3, 14.4 and 14.6.

#### 14.8 Benefit Coverage Participation

#### Staff-Categories

Position Classification	Hours in Normal Work Week			
	Full-time Staff (35 or more)		Part-time Staff (Less than 17.5)	
Regular	Х	Х	Z	
Recurring	Х	Х	Z	
Limited Term	Х	Х	Z	
Contracted	Z	Z	Z	

X - membership in all plans

Z - not eligible for membership in any plans

#### Note:

- 1. Hourly-paid staff members eligible for membership in the Extended Health and Dental Plans, Life Insurance and Trent University Pension Plan have a three-month waiting period before enrolment.
- 2. For recurring employees, the University will contribute 25% of the monthly cost of benefits during layoff, effective September 1, 2004.

# 14.9 <u>Travel Expenses</u>

With previous approval, staff members required to travel as part of their normal University job responsibilities will be reimbursed for reasonable expenses. The administration will publicize and report to the Union details of expense maximums and approval procedures.

#### 14.10 Professional Association

The University shall pay professional and/or licence fees for a staff member who, as a condition of employment, is required to be a member of a professional association or to be licensed.

#### 14.11 Flexible Benefits Plan

The University will provide an annual flexible benefit to each regular or recurring employee equivalent to \$250.00 as of July 1, 2018, \$300.00 as of July 1, 2019, and \$350.00 as of July 1, 2021 per year pro-rated for employees that are employed for less than a year.

The benefit will be applied to members' TrentU Card which can be used to purchase the following:

- Campus parking fees for available lots
- Athletic fees (or Oshawa Civic Centre membership for Durham employees only);
- City Bus Pass;
- Eligible Food Service Providers;
- Any additional services provided by the TrentU card.

Each Flexible Benefit entitlement must be used entirely within the specified year and may not be carried forward to future years. The value of the unused entitlement expires on June 30 of each year.

While the University is currently in receipt of tax advice indicating that a Flexible Benefits Plan confers a taxable benefit on individuals participating in the Plan, it is understood and agreed that in the event the tax status of the Plan, or part thereof, becomes other than is currently understood, the Flexible Benefit, or applicable part thereof, shall be treated in accordance with Revenue Canada regulations.

#### 14.12 Automobile Allowance

For approved automobile travel, employees are entitled to claim mileage allowance, in accordance with prevailing University practices and rates, as amended from time to time.

All staff members using their automobiles for University business should carry sufficient automobile liability insurance. A minimum of \$500,000 is recommended.

"Service Permits" will be provided to employees of the bargaining unit at no cost to the employee, where the employee's duties and responsibilities include a regular and ongoing requirement to use a personal vehicle to transport tools, equipment and/or supplies between various University sites or between the University and locations in the City of Peterborough. Such permits will enable the employee to park in any Red lot or, while performing such duties, in designated University service/loading areas.

Permits will be issued upon authorization of the employee's Department/College Head.

# **ARTICLE 15 - LEAVES OF ABSENCE WITH PAY**

### 15.1 Vacations

Actual dates for vacations must be arranged with, and approved by, the supervisor and Department/College Head. Where scheduling conflicts arise, seniority will be considered by the supervisor when resolving the conflict. Requests for vacation should normally be submitted at least 5 days in advance of the requested days off.

Approval of vacation by the supervisor or department/ college head will normally be given within 5 days of the receipt of the request.

Vacation entitlements are determined according to a staff member's service during the year in accordance with the following schedule:

Years of Service	Vacation Entitlement in Days	Vacation Pay Calculation*	
Less than	One day per month to		
one year	a maximum of ten	4%	
1	10	4%	
2	12	4.8%	
3	15	6%	
4	15	6%	
5	15	6%	
6	16	6.4%	
7	17	6.8%	
8	18	7.2%	
9	20	8%	
10	20	8%	
11	21	8.4%	
12	21	8.4%	
13	22	8.8%	
14	22	8.8%	
15	23	9.2%	
16 - 19	25	10%	
20+	26	10.4%	

\*Less any days taken.

The vacation year runs from July 1 to June 30 and vacation entitlements accrue towards the vacation period beginning July 1 of the succeeding year. However, if the appropriate number of years of service are completed prior to September 30, the additional vacation entitlement accrued in the current year is applicable.

In the event an employee has been employed at the University in previous,

non-bargaining unit position(s) such that their entitlement for vacation under the Employment Standards Act would be greater than their entitlement above, they shall be placed at the appropriate entitlement according to the Employment Standards Act until such a time that they are eligible for increased entitlement based on their OPSEU service.

Normally, vacations may not be carried over from one vacation year to another. However, in certain situations and with prior written approval, vacation days to a maximum of five (5) working days may be carried over to the next vacation year.

Vacation entitlement for part-time staff will be prorated accordingly.

Vacation entitlement does not accrue during:

- (a) any period of layoff, subject to recall;
- (b) any period of unpaid leave of absence (see Articles 16.1, 16.4 and 16.5).

### 15.2 Sickness/Accident/Bereavement During Vacation

When a staff member is hospitalized or bereaved during a vacation period, there shall be no deduction from vacation credits for such period of hospitalization or bereavement entitlement.

A staff member suffering an incapacitating illness or injury which prevents the employee from participating in previously-arranged vacation travel and/or which otherwise seriously prohibits enjoyment of a normal vacation routine, shall receive consideration on a case-by-case basis to extending and/or otherwise re-organizing the vacation period for an appropriate period of time. A medical certificate documenting the illness or injury and the length of time the staff member was incapacitated must be provided. Requests of this nature shall be submitted to the Associate Vice President (Human Resources).

#### 15.3 Holidays

#### Statutory and Non-Statutory Holidays

All regular or recurring full-time or part-time staff members will receive the following statutory holidays with one (1) day's pay, provided they work both the scheduled working day before and the scheduled working day after the holiday:

<u>Statutory</u> New Year's Day Family Day Good Friday Victoria Day Canada Day Non-Statutory Civic Holiday OPSEU Day Labour Day Thanksgiving Day Christmas Day Boxing Day

In addition, all regular and recurring staff members shall be granted the working day before Christmas, three (3) working days between Christmas and New Year's, and one further working day over the Christmas period (to be designated annually by the Department of Human Resources) as paid non-statutory holidays. It is understood that employees temporarily assigned to positions under the provisions of Article 13.14 shall be paid for non-statutory holidays to which they are entitled at the base rate of pay applicable to their regular appointment.

Contracted employees will receive the appropriate statutory or non-statutory holidays, with the exception of OPSEU Day, the working day before Christmas, three working days between Christmas and New Year's, and the further working day over the Christmas period, subject to qualification in accordance with the Employment Standards Act.

Where any of New Year's Day, Canada Day, Christmas Day or Boxing Day fall on either a Saturday or a Sunday, another day (generally the preceding Friday or following Monday) will be declared in its place.

If staff members are required to work on any of the statutory holidays for which they qualify, they will be paid time and one-half for hours worked, plus the compensation for a normal day. If staff members are required to work on any of the non-statutory holidays for which they qualify, they will be paid their normal rate of hours worked, plus compensation for a normal day.

Staff members required to work on a statutory holiday, but who do not qualify for a paid statutory holiday will be paid time and one-half for each hour worked.

A paid holiday on which a staff member does or does not work will not be counted as a normal working day for purposes of calculating overtime entitlements. Overtime entitlement is calculated on hours worked. If you are on vacation over a paid holiday, a further day of vacation or pay in lieu of will be given.

#### 15.4 OPSEU Day (Floating Day)

The University and the Union agree that each regular and recurring staff member is entitled to one other day as an additional paid non-statutory holiday each year. This day will be decided upon by each staff member in consultation with the appropriate supervisor. The OPSEU Day must be used during the Contract year.

#### 15.5 Reporting Absence

Staff who cannot report to work because of sickness or other reasons are expected to telephone within fifteen (15) minutes of their normal starting time, to advise their supervisor of the expected time of their return to work. Staff members who commence work at 16:00 hours or later will make every effort to inform their supervisor(s) of their pending absences as early in the day as possible, and no later than 12:00 hours for the 16:00 hour or 18:00 hour shifts, or 15:00 hours for the midnight shift. However, employees failing to provide notice as stipulated in this Article through circumstances beyond their control shall not be deemed to have violated any of the terms of this Agreement.

Staff should inform their supervisor of the reason for their absence. In the event of illness, exact medical reasons need not be given.

#### 15.6 Medical Certificate

In the event of frequent absences or of an absence of more than four (4) days, a certificate from a licensed medical practitioner which provides satisfactory medical evidence will be required for Department of Human Resources records and to maintain sick leave coverage, if applicable. In an absence of more than two (2) weeks, subsequent doctor's certificates may be required at least once a month in order to support an application for benefits under the Long Term Disability Plan. Subject to Provincial legislation or regulation, these certificates will be at the expense of the employee.

#### 15.7 Sick Leave Policy for Staff Covered Under the Long Term Disability Plan

Each department shall keep accurate individual records of sick leave absences and shall forward appropriate record summaries to the Department of Human Resources each month.

In the event of frequent absences or of an absence of more than four (4) days, a medical certificate will be required. When an absence is prolonged, monthly medical certificates may be required in order to ensure continuation of salary or wages. Any required accommodation, as indicated in the medical documentation described above, will be as per the University Accommodation Policy.

In the event of absence due to sickness or accident and upon receipt of satisfactory medical evidence, a regular or recurring staff member who works seventeen and one-half (17-1/2) hours or more per week and is enrolled in the Long Term Disability Insurance Plan will be paid regular salary or wages up to a maximum of six (6) calendar months. During this period of sick leave, the staff member will continue to assume the cost of the member's portion of benefit contributions.

A staff member who holds a recurring appointment and becomes ill after notice of layoff will still be laid off under the terms of the appointment. Sick leave pay will be paid up to the time of layoff and will start again at the time of recall if the staff member is still ill or disabled, provided a satisfactory medical certificate is produced. The total period of paid sick leave coverage is inclusive of the normal layoff period and will not total more than six (6) months.

If at the end of six (6) months, the staff member is unable to return to work, upon application to the insuring company and the company's approval of the application of disability, Long Term Disability benefits will commence.

Staff members, whose applications for benefits are not approved by the Insurance Company, are entitled to apply to the Employment Insurance Commission, at the time of layoff, for sick benefits calculated on the basis of their insurable earnings. Layoffs resulting from illness or disability are covered under Article 10.14.

#### 15.8 Sick Leave Policy for Staff Not Covered by the Long Term Disability Plan

Regular or recurring staff members who work less than seventeen and one-half (17-1/2) hours per week will be paid their hourly rate at the rate of one (1) normal working day for each month of service dating from the appointment date up to a maximum of sixty (60) normal working days, to the extent that these days have not been used on previous illness.

In the event of frequent absences or of an absence of more than four (4) days, a medical certificate will be required.

When all accumulated sick leave entitlement has been utilized, members shall be laid off in accordance with the provisions of Article 10.14.

Sick leave arrangements for this group of staff members will be outlined in their appointment letters.

# 15.9 <u>Sick Leave Policy for Staff Covered Under the Workplace Safety and</u> Insurance Board

A regular or recurring staff member who suffers an injury at work which is severe enough to warrant loss of time from work will receive full regular salary or wages from the University for a period of up to six (6) months from the date of the absence from work, provided medical evidence satisfactory to the University and the WSIB is received. The regular cost sharing of benefits and pension shall continue as if the staff member was working. The WSIB will reimburse the University at a rate established by WSIB regulations.

If at the end of six (6) months, the staff member is unable to return to work, the University will cease paying salary or wages and the WSIB will be instructed to pay the injured staff member directly. This pay will be calculated at a rate established by WSIB regulations. Coincident with the staff member receiving payment directly from WSIB, an application for long term disability benefits will be made to the insuring company providing these benefits.

If this application for long term disability benefits is approved, the amount of

the WSIB payment will be deducted from the disability benefit payment (See Article 14.3).

At the same time as the WSIB starts to pay the injured or disabled employee directly, the University will commence paying for the total cost of the appropriate benefit and pension plans and will continue to do so while the staff member is eligible to receive WSIB and/or long term disability benefits. Under certain circumstances, disabled and/or injured staff members may engage in rehabilitative employment and continue to receive a portion of their WSIB and/or long term disability benefits.

Staff members, whose applications for WSIB and/or long term disability benefits are not approved, are entitled to apply to the Employment Insurance Commission, at the time of layoff, for sick benefits calculated on the basis of their insurable earnings. Layoffs resulting from illness, injury or disability are covered under Article 10.14.

#### 15.10 Bereavement Leave

An absence, at normal pay, of up to five (5) working days leave shall be authorized in writing by a supervisor for bereavement purposes with a copy of the authorization to the Associate Vice President (Human Resources).

Five (5) working days will be granted in the event of the death in a staff member's immediate family. "Immediate family" shall mean spouse, child, parent, brother, sister, or grandchild. Three (3) working days will be granted in the event of the death of a staff member's father-in-law, mother-in-law, or grandparent. One (1) day absence will be granted for the purpose of attending the funeral of a brother-in-law, sister-in-law, son-in-law, daughter-in-law, aunt, uncle, niece or nephew. For those bereavements involving one (1) or three (3) working days, an additional leave of up to two (2) working days may be granted in exceptional cases, such as where a lengthy trip is required to attend the funeral. Requests for such additional leave shall not be unreasonably denied.

#### 15.11 Civic Leave

Employees subpoenaed to serve as jurors or witnesses in any court shall be granted leave of absence for such purpose, such leave being granted without loss of any part of any of the privileges enjoyed by the employees. Such subpoenaed employees shall present a satisfactory certificate showing the period of such jury service or witness and the amount of compensation received for that purpose. If the employee deposits with the University the full amount of compensation received for jury service or witness service and presents an official receipt therefore, the University shall authorize payment of full salary for the period granted as leave of absence on account of such jury or witness service. If the employee refuses to pay to the University such funds, the individual shall forfeit the right to claim any part of the person's salary for the aforementioned period of absence.

## 15.12 Compassionate Leave and Special Leaves

Leave of absence at normal pay for up to a maximum of six (6) days per year may be granted for compassionate/special reasons. These leaves may be taken as either full or half day increments. Such leaves shall be authorized in writing by the supervisor with a copy to the Associate Vice President (Human Resources). Without limiting the generality of the types of situations where compassionate/special leave may be granted, the following situations will normally qualify:

> Paternity Adoption Illness in the immediate family (spouse, parent, or child) Moving of a household (normally one day for a move within the local Peterborough area) Employee's wedding occurring on a normal workday Major property damage caused by fire or some other personal disaster. Religious Observances Inclement Weather

Although, requests for compassionate/special leave shall not be unreasonably denied, approval will be subject to operational needs.

In the event that the leave request is turned down by the supervisor, the member may appeal the decision to the Associate Vice President (Human Resources).

An individual employee who prefers, in any situation which would give rise to Compassionate or Special leave, to make up time according to Article 12.1 may apply to do so subject to the terms and conditions of that Article. Such make-up time shall be completed in the current pay period or the pay period immediately preceding or immediately subsequent to the time taken off. Make-up time shall not be considered to be overtime, nor shall it give rise to overtime payments. Where an employee makes up time in accordance with these provisions, it is understood and agreed that the University shall not be deemed to have violated any rights under this Article or any other statute or legislation, including the Human Rights Code and Employment Standards Act.

# 15.13 Medical and Dental Appointments

With the permission of their supervisor, employees may be allowed an early departure from or late arrival at work to attend medical and dental appointments. Permission for such absences of no more than two (2) hours per occasion shall not be unreasonably withheld. Early consultation with supervisors on each occasion is required to permit supervisors to organize the workforce.

It is understood that employees will attempt to schedule appointments wherever possible to minimally disrupt normal departmental work schedules.

### 15.14 Maternity/ Pregnancy Leave Sub Plan

If a staff member has been employed at Trent University for one (1) year, and is approved for Employment Insurance Benefits she can file for maternity leave under this sub plan. Under this plan, the Employee will receive either Option A or Option B based on the duration of leave as selected on their Employment Insurance Application:

Option A (12 months): University will pay 95% of the staff member's normal salary or wages for the Employment Insurance waiting period and for the next sixteen (16) weeks, the University will supplement the benefit paid by the Employment Insurance Commission so that when combined with earnings from other sources, the total from both sources will equal 95% of the Trent base salary or wages. The recipient will be required to produce a record of confirmation of receipt of EI benefits from the Employment Insurance Commission before the University can pay the supplement.

Option B (18 months): University will pay 95% of the staff member's normal salary or wages for the Employment Insurance waiting period and for the next sixteen (16) weeks, the University will supplement the benefit paid by the Employment Insurance Commission so that when combined with earnings from other sources, the total from both sources will equal 78% of the Trent base salary or wages. The recipient will be required to produce a record of confirmation of receipt of EI benefits from the Employment Insurance Commission before the University can pay the supplement.

Payments in respect to guaranteed annual remuneration, deferred remuneration or severance pay benefits are not reduced or increased by payments received under this plan.

To be eligible for this benefit, a staff member must have worked at Trent University for twelve (12) months. The staff member must also be planning to return to work no later than eighty (80) weeks from the commencement of combined leaves (maternity and parental) for a period equal to seventeen (17) weeks. An agreement must be signed by the candidate promising to repay the University the University's portion if the staff member decides not to return to work.

During this period of leave, the cost of continuing in the benefit plans will be shared as if the staff member were fully employed.

During maternity leave, vacation credits and seniority accrue in accordance with Articles 10.7 and 15.1.

#### 15.15 Parental Leave Sub Plan

After a staff member has been employed at Trent University for one (1) year,

she/he can apply for a parental leave sub plan. Under this plan all staff eligible under Article 15.14 above may also extend the appropriate sub plan for an additional ten (10) weeks as parental leave providing the parental leave is taken immediately following maternity leave. An additional fifty three (53) weeks of parental leave may also be taken, but shall be considered an unpaid leave and may be eligible for Employment Insurance benefits, but not salary/wages from the University.

To be eligible for this benefit, a staff member must have worked at Trent University for twelve (12) months. The staff member must also be planning to return to work no later than eighty (80) weeks from the commencement of combined leaves (maternity, parental leave and unpaid leave) for a period equal to twenty-seven (27) weeks.

A staff member appointed to a limited term appointment as an external candidate is not entitled to these benefits beyond the completion date of the appointment.

Parental leave benefits for staff not eligible under 15.14 shall be subject to the following conditions:

- (i) If your spouse has not received Employment Insurance maternity leave benefits, the Employment Insurance waiting period - 95% of staff member's normal salary/wages will be paid by the University upon receipt of Employment Insurance acceptance;
- (ii) next eleven (11) weeks the University will supplement Employment Insurance benefits to 95% of normal salary/wages if the 12 month maternity leave option was selected on their application for Employment Insurance, or Option B- the University will supplement Employment Insurance benefits to 78% of normal salary/wages if the 18 month maternity leave option was selected on their application for Employment Insurance;
- (iii) next twenty-five (25) (or fifty-three (53)) weeks are considered unpaid leave and may be eligible for Employment Insurance benefits, but not salary/wages from the University.

The staff member will be required to produce a confirmation of receipt of EI benefits from the Employment Insurance Commission before the University can pay the supplement.

Payments in respect to guaranteed annual remuneration, deferred remuneration or severance pay benefits are not reduced or increased by payments received under this plan.

To be eligible for this benefit, a staff member must have worked at Trent University for twelve (12) months. The staff member must also be planning to return to work no later than sixty three (63) weeks from the commencement of combined parental leave and unpaid leave for a period equal to twelve (12) weeks. An agreement must be signed by the candidate promising to repay the University's portion if the staff member decides not to return to work.

During this period of leave, the cost of continuing in the benefit plans will be shared as if the staff member were fully employed.

During parental leave, vacation credits and seniority accrue in accordance with the Employment Standards Act and Articles 10.7 and 15.1.

A staff member appointed to a limited term appointment as an external candidate is not entitled to these benefits beyond the completion date of the appointment.

#### 15.16 Applications

Leaves under Articles 15.7 through 15.15 are approved by the applicable Department/College Head and reported to Department of Human Resources.

Applications for all other types of leaves with pay shall be made to the Department/College Head who shall refer them to the Department of Human Resources. Such applications will indicate the purpose of the leave and a reasonable outline of the study, research or other activities proposed.

Applications for other types of leave shall normally be submitted at least two (2) months in advance of the proposed starting date, except for extended leaves of absence, for which at least six (6) months' notice shall normally be required.

In the event that an application is turned down by the Department/College Head, the member may appeal the decision with the Associate Vice President (Human Resources).

#### 15.17 Deferred Income Leave

Eligible staff members may fund leaves of absence by deferring a portion of their salary, which is then paid at the time the leave is taken. For more information, see the "Self-funded Leave Plan" booklet, as amended from time to time, or contact the Department of Human Resources.

#### 15.18 Long Service Award

In their 25th year, each member shall have a Long Service Award vacation consisting of an extra five (5) days for a total of thirty-one (31) days' vacation in the 25th year.

# 16.1 Leaves of Absence Without Pay

A staff member may apply for a leave of absence without pay for personal reasons other than illness, providing that the staff member makes a written request stating reasons for the leave to the Department/College Head and subject to the approval of the Associate Vice President (Human Resources). Permission for such leave shall not be unreasonably withheld, subject to workload requirements. Except as provided in Articles 16.2 and 16.3, and subject to the approval of the carrier, Plan provisions, provisions of the *Income Tax Act* and Regulations, and any requirements of the *Pension Benefits Act* R.S.O. 1990 and Regulations, an employee may, at their option and upon written application to the Department of Human Resources, continue participation in benefit and pension plans, provided the employee pays the full premiums cost(s), including any share normally contributed by the University.

Current provisions permit a member on unpaid leave to continue long term disability coverage for a maximum of four (4) months; pension contributions during unpaid leave are limited to a life-time maximum of five (5) years.

# 16.2 Pregnancy Leave

A staff member is entitled to a flexible seventeen (17) week unpaid leave of absence for pregnancy provided the member has a minimum of thirteen (13) weeks of continuous service prior to the expected date of delivery as indicated by her medical certificate. The member is required to give two (2) weeks written notice of the date on which she intends to begin her leave. All members may begin their leave at any time within seventeen (17) weeks before the expected date of delivery.

If a staff member is required to be absent for a period longer than seventeen (17) weeks as certified by a legally qualified medical practitioner, or if her work is materially affected by her pregnancy, and the University requires an extension of leave, any further leave period is covered under the University's sick leave policy.

The staff member is entitled to not less than six (6) weeks post-natal leave after the date of birth. The Employment Standards Act insures the right of the employee to return to the same or a comparable job without loss of seniority or benefits or drop in salary, if pregnancy leave of not longer than seventeen (17) weeks is taken.

During pregnancy leave, the cost of continuing membership in the University's benefits plans shall continue as if the member were actively at work. In the case of benefits where costs are shared (i.e., pension, life insurance), if the staff members are willing to pay their normal share of the cost, the University will contribute its portion of the costs.

During pregnancy leave, the vacation credits do accrue in accordance with Article 15.1.

If the staff member returns to work on a regular basis the period of pregnancy leave is considered time worked for seniority purposes.

### 16.3 Parental Leave

Under the terms of the Employment Standards Act, all employees who are parents (both natural and adoptive) are entitled to sixty-three (63) weeks of unpaid parental leave, provided they have been employed for a minimum of thirteen (13) weeks prior to the estimated date of delivery or adoption of a child. For natural mothers, parental leave must commence at the end of maternity leave. For others, the leave must commence no later than fifty-two (52) weeks after the date of birth or adoption.

Natural mothers are entitled to seventeen (17) weeks of maternity leave plus sixty-one (61) weeks of parental leave. Fathers and Adoptive parents are entitled to sixty-three (63) weeks of parental leave.

Under the terms of the Employment Insurance Act, parental benefits are available to eligible parents (both natural and adoptive) for a total of sixtythree (63) weeks per family. The thirty-five (35) weeks can be used by one parent or split between the two. For individuals not previously in receipt of maternity benefits, there is a two-week waiting period during which parental benefits will not be paid.

During parental leave, the cost of continuing membership in the University's benefits plans shall continue as if the member were actively at work. In the case of benefits where costs are shared (i.e., pension, life insurance), if the staff members are willing to pay their normal share of the cost, the University will contribute its portion of the costs.

During parental leave, the vacation credits do accrue in accordance with Article 15.1.

# 16.4 Long-term Leave of Absence

A long-term leave of absence may be requested for academic or professional study reasons, and may extend from four (4) to eighteen (18) months. While approval for such leave shall not be unreasonably withheld, staff members should not expect to necessarily return to the same position at the end of this leave. Where possible, a contract replacement may be appointed for the period of the regular staff member's leave.

# 16.5 Short-term Leave of Absence

Short-term leave of absence may be requested for personal reasons, and may extend up to four (4) months. Approval for such leave shall not be unreasonably withheld, subject to workload requirements. Staff members

may expect to return to the same position at the end of this leave. Any replacement approved shall be a contract appointment.

## 17.1 Job Descriptions

The University will provide each employee covered by the Agreement with a general Job Description, based on the job questionnaires submitted to the Department of Human Resources, including duties, responsibilities, qualifications, and classification for his/her job. The Department of Human Resources has the sole responsibility and authority for the format, content and final approval of Job Descriptions, and for any amendments or modification thereto.

# 17.2 Job Questionnaires

Upon request, employees will have access to the job questionnaire used for the purpose of evaluating their position as well as the assigned factor scores used to establish the job band.

# 17.3 Joint Job Evaluation System

# <u>General</u>

Jobs shall be classified and wages shall be paid during the term of this Agreement, in accordance with Schedules A and B of this Agreement, and with the Joint Job Evaluation System, copies available in the Department of Human Resources. It is understood that the Trent University and OPSEU Joint Job Evaluation System, implemented February 1, 2001 (factor definition, rating criteria, point chart and joint committee evaluation), forms part of the Collective Agreement. All positions are deemed to be properly evaluated as of February 1, 2001.

# 17.4 The Joint Job Evaluation Committee

- (a) A Joint Job Evaluation Committee, a subcommittee of joint committee Article 6, will be formed to maintain and evaluate the classification of positions. Members of the JJEC shall be entitled to make reasonable adjustments of their normal duties as per Article 4.10.
- (b) The Joint Job Evaluation Committee will consist of 6 representatives. The committee will consist of two experienced Job Evaluation Committee members from the previous committee, one from OPSEU and one from the Department of Human Resources, plus two (2) employees elected by the Union and two (2) employees, managers of OPSEU staff, appointed by the Employer. Committee membership will consist of a two-year rotating term, for all JJEC members, one as new and one as experienced.
- (c) Training will be provided by the two experienced Job Evaluation Committee members from the previous committee, one from OPSEU

and one from the Department of Human Resources.

- (d) All Committee decisions will be by consensus; however, two (2) Union representatives plus two (2) Employer representatives are required for a quorum. Any decisions will require an equal number of JJEC representatives from both parties be present.
- (e) This committee will be scheduled to meet on a regular basis (monthly), with frequency dictated by the number of reviews outstanding, the meeting being cancelled in the absence of any evaluations to be processed.

# 17.5 Joint Working Sub-Committee

In order to avoid lengthy delays in job posting, a sub-committee will be formed to evaluate new or substantively-altered positions prior to posting. This Sub Committee will consist of the two (2) experienced members, one Employer representative and one Union representative. The Working Sub-Committee will normally render their recommendations within five (5) working days from receipt of the completed job questionnaire to the Human Resources Department.

# 17.6 New Jobs

The University may create new jobs at any time during the term of this Agreement. A job description for the new position will be completed by the manager or supervisor and submitted to the Joint Working Sub-Committee for their review. Within five (5) working days this Sub-Committee will exchange recommendations and if in agreement on the evaluation of the new position, the new position is deemed to have been evaluated and may be posted in keeping with Article 18, including the job band assigned.

The employer will include the banding of any new positions in the Master Job Evaluation Spreadsheet, to be distributed to JJEC monthly.

If there is an objection, the position will be reviewed at the next monthly meeting of the larger Joint Job Evaluation Committee. Should an employee bid into a position which is subsequently increased in band, the employee will be paid at the higher band level retroactive to the commencement of their appointment to the position. If the band is subsequently decreased, as a result of the JJEC review, the employee's rate will be reduced effective the date of the decision of the Joint Job Evaluation Committee and the employee may exercise their rights under Article 13.11.

When a vacancy occurs in a multiple incumbent position and the supervisor elects to substantively alter the position prior to posting such that the JJEC Sub-Committee upgrades the position, those members with more than twelve (12) months of service, who have been determined by the supervisor to be both in the substantively altered position and qualified, will be upgraded accordingly, effective the date that the vacant position is submitted to Human Resources for banding and posting.

#### 17.7 Substantively-altered Jobs

The University may substantively alter existing jobs at any time during the term of this Agreement.

An incumbent in a position which has been altered may request a review of the job evaluation based on substantive change of a permanent nature which they believe will change the banding of the position.

The Union or the Employer may also request a review of the job evaluation based on a substantive change of a permanent nature, which they believe will change the banding of the position.

Submissions must be signed by the party requesting the review and a copy provided to the employee, the supervisor and the Department of Human Resources.

The party requesting a review will submit their request in writing to the Joint Job Evaluation Committee, c/o Department of Human Resources. The written request should be in the form of a full job evaluation submission such as: a new completed questionnaire, a revised job description or a letter which includes a description of how the position has changed from the current questionnaire. The preferred submission format is an updated job description, with clear identification of changes and an accompanying letter, explaining how the position has changed since it was last evaluated. This document will be submitted to the Joint Job Evaluation Committee for their review at their next monthly meeting.

Should the committee agree on a change in a job evaluation, the effective date of such change will be the date that the request for review is received in the Department of Human Resources.

The decision of the Joint Job Evaluation Committee will be communicated by the Department of Human Resources to the employee and the Local President, or designate, within fifteen (15) days of that meeting.

#### 17.8 Mediation/Arbitration

If the members of the Joint Job Evaluation Committee fail to reach agreement on the scoring and banding of a position, the issue may be submitted to Mediation/Arbitration. A mediator/ arbitrator jointly agreed by the parties will be asked to pre book one Med/Arb day per quarter to hear issues between the parties, and if required render a decision.

It is understood and agreed that the arbitrator's only jurisdiction shall be with respect to a determination of classifications or rates of pay, and shall be expressly restricted to the existing job classifications and job rates as set out in Schedules A and B of this Agreement, and to a determination of whether the grievor's job is properly classified.

# 17.9 Appeals of Classification Reviews

Within six (6) months of the meeting date at which JJEC's decision was rendered under Article 17.7, an employee who claims his/her assigned job is improperly classified and that he/she should be properly classified to another classification named in Schedule B, may appeal the committee's findings by presenting a request for review in writing to the Joint Job Evaluation Committee, c/o the Department of Human Resources. In the event that the appeals deadline falls on a non-working day, the deadline shall be considered to be the next working day.

The written request must specify the classification claimed by the employee to be inappropriate together with those individual job factors which are being contested and the substantive basis on which the claim is advanced. Any changes from the original questionnaire should be included. The employee may also request to give a brief verbal submission to the Joint Job Evaluation Committee on the date of the joint review.

The Committee will review the whole job, not just individual factors. Should the Committee make a change to a job factor, all other comparator positions will be reviewed to ensure pay equity compliance. The decision of the Joint Job Evaluation Committee will be communicated by the Department of Human Resources to the employee and the Local President, or designate, within fifteen (15) days of that meeting. The Joint Job Evaluation Committee shall be under no obligation to accept or respond to more than one (1) request for review of the same position within any twelve (12) month period.

Should the committee agree on a change in a job evaluation, the effective date of such change will be the date that the request for review is received in Department of Human Resources.

## 17.10 Band Increase

- (a) Employees reclassified to a higher band level shall be placed at a step level in the new band that represents as a minimum a one (1) step increase from the previous salary.
- (b) It is understood that there shall be no retroactivity payment, including any retroactivity awarded through arbitration, prior to the date of the receipt of the written request for review in Department of Human Resources.

#### 17.11 Grievance of Classification Reviews

(a) An employee who is in a position which has been subject to the procedures outlined in 17.9 above, and who is not satisfied with the decision of the Joint Job Evaluation Committee, may file a grievance directly in writing, within two (2) months of the meeting date at which JJEC's appeal decision was rendered.

- i. Once an employee receives and is not in agreement with the decision of the Joint Job Evaluation Committee, he/she may file a grievance on a standard OPSEU local 365 grievance form.
- ii. The employee will have two (2) months from the meeting date at which JJEC's decision was rendered to file a grievance.
- iii. The grievance shall be limited to the consideration of those relevant facts submitted for review to the Joint Job Evaluation Committee.
- iv. The grievance should be limited to the employee's perceived misapplication of the Tool by the JJEC, and should not be a denigration of the joint process itself.
- v. Only those factors that are being contested, and the substantive basis on which the claim is advanced should be included.
- vi. The grievance will be initiated at Step 3 of the grievance process.
- vii. If the parties are unable to resolve the grievance at Step 3, it will be referred to arbitration, to an Arbitrator mutually agreed upon by the parties.
- (b) Referral to Arbitration
  - i. A written submission must be submitted by the member in consultation with OPSEU to OPSEU and copied to Human Resources within 30 working days of the decision of the parties to refer the grievance to arbitration.
  - ii. An Employer representative will subsequently prepare a written submission explaining the decision of JJEC. This will be limited to those facts reviewed by the Committee, including any verbal submissions by the employee or their supervisor. OPSEU local 365 will be copied on the submission to the Arbitrator.
- (c) The arbitrator may issue a decision in writing, or by meeting with the parties at the next quarterly meeting, or at their earliest available date acceptable to the parties. A brief written notice of his/her decision will be issued within ten (10) working days of the hearing or upon submission in writing. The parties may jointly decide to forgo a meeting with the Mediator/Arbitrator and refer the written submissions to the Arbitrator for decision.
  - i. The grievor will be make their submission first, and will be given the opportunity to make a 30-minute verbal submission to the Arbitrator. The Chief Steward and/or the Union Steward from their area will present argument during the process. It is further understood that the Regional OPSEU Representative may attend so that he/she may oversee the proceedings to ensure bargaining members' rights and the integrity of the joint process.

- ii. The experienced JJEC Employer representative will respond with the experienced JJEC Union representative in attendance.
- (d) It is anticipated that the arbitrator will be able to hear four (4) cases per quarterly meeting. If at the time of the regularly scheduled meeting, the number of classification grievances outstanding is less than four, the cases will be deferred until the next meeting date, with no one single case waiting longer than a 6-month period.

### 17.12 Cost Sharing

The University and the Union shall each pay one-half (1/2) of the remuneration and expenses of the Mediator.

Guidance on the Trent University OPSEU Job Evaluation process can be found on the Trent University Human Resources compensation webpage.

# ARTICLE 18 - JOB POSTING PROCEDURES

# 18.1 Job Postings

- Permanent regular or recurring vacancies, or new regular or recurring positions, shall be posted on designated bulletin boards and on the Trent University web site for eight (8) working days from the date of issue. E-mail copies will be sent to locations off the main campus and the OPSEU office. In addition, hard copies of job postings shall be placed on designated bulletin boards.
- 2. Contract positions less than three (3) months in duration are not subject to the job posting or job competition process. Contracts that exceed three (3) months in duration, and contracts with an unknown end date shall be posted on the bulletin boards for the information of all staff.
- 3. All postings shall include:
  - (a) title and description of the position;
  - (b) position supervisor;
  - (c) salary classification and salary range (as published);
  - (d) position definition including the schedule for recurring positions;
  - (e) required qualifications including education, experience and special skills;
  - (f) deadline for receipt of applications;
  - (g) starting date.
- 4. A current list of bargaining unit vacancies shall be provided quarterly to the Union one-week in advance of each Joint Committee meeting. The list shall include, but not necessarily be limited to, job title, job number, date the vacancy began, reason for job vacancy, and status of the posting and filling of the vacancy. Should a new computer software program be able to produce monthly reports, then these shall be provided to the Union on a monthly basis.

## 18.2 Applications

Applications may be submitted in writing or electronically to Human Resources prior to the deadline. In the case of regular/recurring positions the University will consider regular and/or recurring, internal applicants, and shall not consider and/or appoint an external candidate where an internal applicant possesses the required qualifications for the position including education, experience and special skills.

Members who have been laid off as per Article 11.6 or who are on the recall list shall be considered prior to a position being made available to other internal applicants, and qualifications are expanded as described in Article 11.6.

Those internal applicants who, in the opinion of the Hiring Committee, lack the required qualifications will be notified in writing, and upon request to Department of Human Resources, will be advised of the areas in which their qualifications do not meet the posted requirements.

The University shall be under no obligation to consider applications from, or to select, employees who have been hired, promoted or who have successfully applied for a transfer within the previous six (6) months. The University shall be under no obligation to select any employee who has made application.

Department of Human Resources will forward applications of internal applicants to the hiring committee and the committee will review these together with:

- (a) the applicant's previous work record at Trent, including reports from former and current supervisors;
- (b) seniority;
- (c) known relevant experience outside Trent;
- (d) test scores;
- (e) known particulars of qualifications.

In the event there are no successful internal applicants, the University may consider applications from external and contract employees.

#### 18.3 Selection

In selecting an employee to fill a permanent regular or recurring vacancy or a new regular or recurring position, the University shall consider the applicant's overall qualifications, including education, experience and special skills, and where these factors are relatively equal, then seniority would be the deciding factor. Relatively equal shall mean a range of 10 percentage points on the selection evaluation scale.

The weighting of the criteria utilized in the Selection Evaluation process shall be:

Application	5 points
Training and Experience	30 points
Interview	45 points
Assessment of Performance in Current/Past Position(s)	20 points

Department of Human Resources will notify both successful and unsuccessful internal candidates. In addition, unsuccessful candidates will be advised of their right to receive in writing upon request to Department of Human Resources, the reasons for the decision.

#### 18.4 Release from Present Position

Successful applicants should be released from their present position as soon as possible, but normally at least within twenty (20) working days of notification of their selection. During any overlap period, and as scheduled by the Department of Human Resources, the successful applicant shall be expected to spend some time in the new position receiving orientation, as well as assisting with the orientation of a replacement for the position being vacated.

# **ARTICLE 19 - CAREER DEVELOPMENT**

19.1 Trent University is committed to assisting staff in career planning and in seeking opportunities for career growth.

Any regular or recurring employee who wishes to establish a formal career plan may apply through Human Resources. The University will provide the services of an experienced career counsellor who will meet with the employee and in a confidential capacity assist the employee through testing and interviews to prepare a future plan.

Development of a formal plan, which shall include further education or skill development through a counsellor is limited to fifteen (15) staff per year. If, at any time through the year, there are more applicants than spaces remaining in the program, space will be allocated on the basis of seniority, and unsuccessful applicants will be advised to apply the following year.

Once a formal plan has been developed, the employee shall meet with Human Resources to provide a written copy of their formal career plan. The employee together with Human Resources will seek opportunities which are appropriate for development and which fit within the individual's agreed-upon career plan. Such opportunities may be found for example in anticipated or vacant contracted positions, limited term appointments, an opportunity identified in the performance assessment process, a secondment outside of the bargaining unit or external to Trent University. While the staff member is on secondment outside of the bargaining unit, he/she will continue to accrue seniority up to a maximum of twelve (12) months. The Parties may agree to extend the timelines. If the employee does not return to their position within twelve (12) months, or any extended period, he/she will be deemed to be terminated from the bargaining unit and their home position will be posted. Should more than one qualified employee indicate an interest in the same contracted opportunity, selection will be by seniority.

During the time that an employee is in a developmental assignment, their home position may be filled by another employee on a contract, subject to the provisions of Article 10.1(c) and 13.14.

Participants in Career Development opportunities may be considered for a position for which he/she does not meet all of the qualifications, but who has been identified as being appropriate for such role. An applicant under this program will have priority consideration for temporary positions in the OPSEU bargaining unit at Trent, providing there are no qualified employees on layoff. Such employee must compete for any regular or recurring positions.

Compensation will be at the rate of pay assigned to the Career Development opportunity, not the incumbent's home position.

Should the employee elect to enter the program but does not complete the program or does not submit a formal plan within a twenty four month time period, the University will have no further obligation to the employee, nor is

the employee entitled to priority consideration for developmental positions.

A maximum of two priority consideration career development opportunities will be allowed for each employee in the Career Development Program.

An employee will remain in the program until the earlier of either receiving two (2) priority consideration career development opportunities, or five (5) years following the completion of their career development plan.

Employees shall be eligible for re-entry to the career development subject to available space above, after a period of 5 years has passed since they were last enrolled.

New employees shall have a waiting period of three years before they can enrol in the career development program.

The University shall supply to the Union on a biannual basis information listing the names of individuals enrolled in the plan and their status and progression through the plan.

The University should be entitled to recoup any training costs should an employee depart from the University within 120 days of completing training.

#### 19.2 Academic Courses and Professional Study

Staff members may, on the recommendation of their Department/College Head, and at the discretion of the Associate Vice President (Human Resources) or designate, be given the opportunity and may be required to participate in seminars, workshops, and short courses to keep up to date with the knowledge and skills of their respective fields.

The staff member who attends a conference or convention at the request of the University to represent the interest of Trent University, or who is required to attend a course for the purposes of intensive retraining or upgrading, shall be eligible for full pay, travel expenses, and reimbursement of registration fees.

The staff member whose participation in a conference or convention in an official capacity (e.g. present an address, or give a course in a field of specialization), as approved by the supervisor and Department/College Head, is eligible for leave with pay and reimbursement of registration fees and travel expenses.

A joint University/Union committee will be established to meet on a quarterly basis to make recommendations on training courses which would benefit a group of employees in their career development. Courses may be in areas such as developing technology, software applications, report writing skills, problem solving techniques, statistical analysis or business programs. From the recommendations of the committee, a training course, preferably one granting a certificate or in conjunction with a degree program, will be held on campus each Fall and Winter term. Courses will be arranged in conjunction with the timing best suited to the participants, for example over a lunch period or coincident with the end of shift, or opportunity will be provided to make up loss of regular hours of work.

### 19.3 Education Leave

The University encourages staff to work towards self-development by enrolment in academic and technical courses offered by secondary and post-secondary institutions.

# 19.4 <u>Policy Respecting Rebate of Tuition Course Fees to Eligible University</u> <u>Staff Members</u>

- 1. An eligible University staff member is one who holds a regular or recurring appointment at Trent University.
- 2. For purposes of this policy, a course is defined as a program of formal instruction offered by a secondary or post-secondary institution in the Province of Ontario, up to and including graduate level programs, the successful completion of which, in the opinion of the supervisor and the Associate Vice President (Human Resources), will be of benefit to the staff member in the performance of job duties at the University and will result in some educational credit being obtained.
- 3. Eligible staff applying to take courses must be prepared to meet normal admission requirements and to comply with published deadlines, etc.
- 4. Tuition course fee rebates shall be limited to the equivalent of one (1) undergraduate credit between September and April and one (1) undergraduate credit between May and August. The maximum tuition fee rebate for all Graduate Programs is the domestic part time graduate program fee per term, not the Masters of Education or future Professional Master's program rates. Where the hours of instruction fall within working hours, the employee shall be required to make up for any time lost by taking the course. Where there are multiple offerings to the course, the employee is expected to select the timing so as to provide minimal disruption to the work unit.
- 5. When staff members are required to write examinations during working hours for courses taken under the tuition rebate policy, this time away from work will be considered time off with pay. Staff members will not be required to make up time lost from work while writing scheduled examinations. Staff members will inform their supervisors of the scheduled examination dates as early as possible.
- 6. The University will waive the fee for one undergraduate credit taken between September and April and one undergraduate credit taken

between May and August at Trent University, or one domestic part time graduate program fee (not the Master's of Education or future professional Master's program rates) per term, or rebate the fee for a course taken at another academic institution provided written authorization for both the taking of the course and the waiving or rebating of the fee has been obtained from the supervisor prior to the commencement of the course. Proof of successful completion will be required to receive a rebate or to validate the fee waiver.

- 7. The forms for written authorization should be obtained from the Department of Human Resources. This Department should be contacted if authorization appears to be improperly withheld.
- 8. Failure to successfully complete a course will result in the staff member being held financially responsible for the full course fee. Where there has been a waiver, the charge will be placed on the staff member's account. Where the staff member has withdrawn from a Trent course without academic penalty, the appropriate fee up to the date of withdrawal in accordance with Trent University policy will be charged to the staff member's account.
- 9. Ancillary fees consisting of the College fee, Health Service fee, and Athletics fee will be waived. Staff wishing to use the Athletics complex will be required to purchase a membership.
- 10. The University will not be financially responsible for fees for staff who only wish to audit a course.
- 11. Where the course is taken at an accredited university other than Trent, the fee rebate will be limited to Trent's undergraduate course fee or domestic part time graduate program fee, exclusive of ancillary fees.

#### 19.5 Dependent's Tuition Waiver

Dependents of a staff member who holds a regular or recurring appointment, enrolled in courses at Trent University, shall have their normal undergraduate academic fees waived for the duration of this Agreement. Ancillary fees are not waived. Admission and progression standards for such students shall be those applicable to other students.

Tuition waivers must be submitted no later than 30 days before the commencement of the term. The receipt of a tuition waiver may impact the eligibility for federal or provincial initiatives.

#### 19.6 Definitions of Dependency

Dependency shall be considered to apply to anyone living in the family relationship of husband and wife (including common law) or parent and child.

Spouses: In any given academic year, tuition will be waived for one spouse

only for each member of the bargaining unit.

<u>Children</u>: The applicable definition of children shall include adopted children, foster children, and married children, subject only to the condition that the children in question be twenty-five (25) years of age or under.

### 19.7 On-the-Job Training

The parties recognize the desirability of staff development through on-the-job training. The University agrees to consult with the Union on training and development needs of staff members and the ability of the University to meet these needs during regular working periods without affecting the salary/hourly rate of the staff members concerned.

### 19.8 University Committees

Staff members accepting an office or committee work with a University committee must first obtain the authorization of their supervisor and/or Department/College Head. Such authorization shall not be unreasonably withheld. The University recognizes the contribution of staff members who become involved in such committee work, and wherever possible, the University will provide time off during working hours for activities which are an integral part of the operation of such committees, and at the request of the employee and with the approval of the supervisor, time off in lieu for committee meetings scheduled outside normal working hours. Time is normally granted to attend meetings during normal working hours, but must be approved by the supervisor. Committee members should notify fellow departmental staff members to ensure that service is maintained in departments at all times. Costs for committee use of reasonable quantities of supplies and services (e.g. stationery, photocopying, postage, telex, IUTS) will be absorbed by the University. Appropriate budget accounts should be employed for the charging of such materials or services.

# **ARTICLE 20 - GRIEVANCE AND ARBITRATION**

### 20.1 <u>Grievance Definition</u>

A grievance shall be defined as any dispute or difference arising out of the interpretation, application, administration or alleged violation of the provisions of this Agreement. No grievance shall be deemed to be invalid or abandoned due to a minor technical irregularity.

### 20.2 Types of Grievance

- (a) An individual grievance is a grievance which involves a single individual.
- (b) A group grievance is one which involves two (2) or more employees. Such a grievance may be commenced as a group grievance, or similar individual grievances seeking a common redress may be consolidated as a group grievance.
- (c) A policy grievance is one involving a question of general application, administration or interpretation of this Agreement.

### 20.3 Carriage of Grievances

An individual employee shall have the right to initiate a grievance at the informal complaint stage (see 20.4, below), provided however that any such grievance may not be advanced to the Formal Steps without the Union being a signatory to the written grievance.

#### 20.4 Informal Complaint Stage

Before the formal grievance procedure is initiated, efforts will normally be made to resolve the dispute through informal discussion with the immediate supervisor. In such cases, complaints shall be discussed with the immediate supervisor within ten (10) working days of the incident or original circumstances giving rise to the complaint, or within ten (10) working days after the complainant became aware of or ought reasonably to have known of the incident or original circumstances giving rise to the company the complaint. An authorized Union representative may accompany the complainant at the informal stage, if the complainant so wishes. The immediate supervisor shall respond within five (5) working days after the complaint was discussed and, if not satisfied, the complainant may file an individual grievance as set out in Article 20.5.

## 20.5 Formal Grievance Steps

An earnest effort shall be made to settle grievances fairly and promptly as set out below:

<u>Step 1</u>

Failing resolution of the complaint at the Informal Stage, a formal grievance may be filed. If so, it shall be set forth in writing, on a form provided by OPSEU and mutually agreed by OPSEU and Department of Human Resources, and signed by the grievor and authorized OPSEU representative. The written grievance shall set forth the details of the grievance, a statement of the matter(s) in dispute, the provision(s) or interpretation of the Agreement that has allegedly been violated and the remedy sought. The written grievance shall be given to the grievor's immediate supervisor within thirty (30) working days after the incident or original circumstances giving rise to the complaint, or within thirty (30) working days after the grievor became aware of or ought reasonably to have known of the incident or original circumstances giving rise to the complaint. A copy of the written grievance shall be forwarded to the Department/College Head and Department of Human Resources. The immediate supervisor will reply, in writing, within ten (10) working days following receipt of the written grievance.

#### <u>Step 2</u>

If the grievance is not settled at Step 1, it shall be presented to the Department/College Head concerned within five (5) working days following receipt of the reply as set forth in Step 1 above. The Department/College Head will give a written reply within a further five (5) working days.

#### Step 3

If the grievance is not settled at Step 2, it shall be presented to the Associate Vice President (Human Resources) within five (5) working days following receipt of the reply as set forth in Step 2 above. Within five (5) working days of receiving the written grievance, the Department of Human Resources shall convene a meeting of the concerned parties to the grievance. After this meeting, a written reply will be given within five (5) working days.

#### Step 4

If the grievance is not settled or otherwise resolved at Step 3, the grievance may be referred to Arbitration (see Articles 20.7 to 20.13 below).

#### 20.6 General Provisions

- (a) Individual grievances shall be initiated at Step 1; Group grievances may be initiated at Step 2, if all the employees concerned are employed in the same department, or at Step 3, if employed in different departments; Policy grievances may be filed at Step 3.
- (b) Notwithstanding sub-paragraph 20.6(a) (above), any grievance involving discipline, suspension or discharge may be commenced at Step 3 of the Grievance Procedure. In addition, where an employee grieves an internal job competition for a position outside his/her department, the grievance may be commenced at Step 3 of the

Grievance Procedure, provided that a copy of the grievance shall also be submitted to the employee's immediate supervisor.

- (c) In the case of a University Policy Grievance which is submitted at Step 3, the grievance shall be submitted to the OPSEU President, or designate, and the referral of a University Policy Grievance to Arbitration shall be made by written notice to the OPSEU President, or designate. In such cases, the stated time limits and requirements for meetings and responses shall apply, as specified, with the grieving and responding roles appropriately reversed.
- (d) All agreements reached under the formal steps of Grievance Procedure (i.e., Steps 1, 2, or 3), between the representatives of the parties, including any agreement as to whether the settlement of a particular grievance shall stand as a precedent for other cases, will be final and binding upon the parties and the employees. Any such settlements shall be reduced to writing, signed by the authorized representative of both parties, and copied to the authorized Union representative, the grievor and the Department of Human Resources.
- (e) The parties agree to follow the grievance steps as outlined above, in Articles 20.4 and 20.5, in accordance with the time limits and conditions contained therein. If at any step, the responding party fails to give a written reply within the required time limit, the grieving party may advance the grievance to the next step at the expiration of the time limit. If the grieving party does not advance the grievance to the next step within five (5) working days following receipt of the written reply, that particular grievance shall be deemed to be settled on the basis of the last reply. If dissatisfied with the decision at any step in the Grievance Procedure, the grieving party may take the grievance to the next step.
- (f) The University acknowledges that employees have the right to Union assistance in preparing and presenting a complaint and/or grievance, including any complaints or grievances in relation to disciplinary matters. The Union agrees to advise the University, on a current basis, of the names of its authorized grievance representatives.
- (g) The University recognizes the role of authorized Union representatives in investigating disputes and assisting/participating in the grievance and arbitration process. As much as possible, investigation, consultation and participation in grievance matters shall be scheduled in such a way as to minimize conflict with assigned duties. Where that is not possible, grievors and authorized Union representatives will be entitled to reasonable time off, without loss of normal pay, benefits or seniority, for the purpose of attending grievance step meetings with the employer or arbitration hearings. In such cases, employees shall not leave work without having obtained permission from their immediate supervisor. Such permission shall not be unreasonably withheld.

- (h) "Working Day", for the purposes of these grievance and arbitration provisions, excludes Saturdays, Sundays and Holidays (see 15.3).
- (i) Subject only to availability through normal scheduling, the University shall provide suitable meeting rooms for the purpose of grievance investigations, discussions, and meetings.
- (j) The time limits specified in both the Grievance and Arbitration Procedures may be extended only by written consent of both parties to the Agreement. Similarly, any step of the Grievance Procedure may be waived by written consent of both parties.

#### 20.7 Referral of Grievance to Arbitration

Within ten (10) working days of receipt of the Step 3 reply, the grieving party shall give notice, in writing, of its intention to refer the grievance to arbitration. Such notice shall also specify the name of the party's nominee to the Arbitration Board. Within ten (10) working days of receipt of the written notice to refer the grievance to arbitration, the responding party will advise, in writing, of the name of its nominee.

#### 20.8 Selection of Chairperson

The two (2) nominees to the Arbitration Board shall select a Chairperson within ten (10) working days of the appointment of the latter of them.

If the nominees fail to agree within the specified period, either party may request in writing, with a copy to the other party, that the appointment be made by the Office of Arbitration, Ministry of Labour, for the Province of Ontario.

#### 20.9 Appointment of a Single Mediator/Arbitrator

Notwithstanding the previous provisions governing the establishment of a three-person Arbitration Board, the parties may, by mutual agreement, refer the grievance for determination by a single arbitrator.

Such single arbitrator may be asked by the Parties, at date of appointment or at the arbitration hearing, to first attempt to mediate a resolution of the issue and then, if unsuccessful, to arbitrate.

#### 20.10 Exclusions

No person may be appointed to an Arbitration Board who has been involved in an attempt to negotiate or settle the grievance, or who has a conflict of interest because of a personal involvement in the subject matter of a dispute.

#### 20.11 Jurisdiction

The Arbitration Board shall have no jurisdiction or authority to amend, alter, modify or add to any of the provisions of this Agreement, or to substitute any

new provisions in lieu thereof, nor to give any decision inconsistent with the express terms and conditions of this Agreement. This provision does not affect the Arbitration Board's authority to modify disciplinary penalties. Except where otherwise expressly provided in this Article 20, the powers of the Arbitration Board shall be in accordance with those specified in the Ontario Labour Relations Act.

### 20.12 Fees and Expenses

Each of the parties hereto will bear the fees and expenses of the nominee appointed by it, as well as all costs related to the presentation of its own case, and the parties will share equally the fees and expenses of the Chairperson of the Arbitration Board, including the costs of the hearing room. The parties agree to use University facilities wherever possible.

### 20.13 Decision of the Board

- (a) The decision of the majority shall be the decision of the Board. Where there is no majority decision, the decision of the Chairperson shall be the decision of the Board.
- (b) The decision of the Board is binding upon the parties and upon the employees covered by the Agreement who are affected by the decision.

# **ARTICLE 21- ANIMAL CARE WORKERS**

The parties agree that there are standards defined by the Animals for Research Act and Canadian Council on Animal Care and included in the definition of standards is a recommendation for provision of services during a strike or lockout. Without such assurance that Research Animals are under proper care in the event of a strike or lockout, Trent University would be found to be non-compliant.

For this purpose, the parties agree to discuss a protocol for Animal Care staff prior to any strike or lockout.

# **APPENDIX I- Pension Health Care Allowance**

# MEMORANDUM OF UNDERSTANDING

### Re: Pension Plan – Health Care Allowance

Note: Appendix II has been included in the Collective Agreement for purposes of reference.

WHEREAS The Retiree Health Care Allowance, as described in Section 7 of the Memorandum of Agreement with Respect to Pension Plan and Retiree Health Care Reimbursements, as provided in Appendix VI to the collective agreement for the period July 1, 2007 to June 30, 2010, applies to members who elect to take early retirement from July 1, 2007;

AND WHEREAS previous collective agreements have provided for a Retiree Health Care Allowance with application to specific retiree groups. As these are ongoing benefits for staff members who retired prior to July 1, 2007, the terms of these arrangements are summarized below:

The parties agree to the following summary of previous Collective Agreements which are included for purposes of reference:

1. A Health Care Allowance for existing retirees who were drawing pensions on June 30, 1998 was implemented. This non-bankable fund reimburses retirees and survivors for eligible health-related expenses, including health plan premiums as defined in *The Income Tax Act*. It is created annually for each retiree or survivor based on one of the following formulae:

i) for retirement after June 30, 1987, \$25 for each year of Pension Plan membership, to a maximum of \$500;

ii) for retirement before July 1, 1987, \$45 for each year of Pension Plan membership, to a maximum of \$1,000;

iii) deferred pensioners are ineligible to receive a health account.

(Source Document: Appendix II, Pension Plan Framework Agreement, Collective Agreement between Trent University and OPSEU, July 1, 1998 to June 30, 2001; and duplicated in Appendix I, Pension Plan Framework Agreement, Collective Agreement between Trent University and OPSEU, July 1, 2001 to June 30, 2004 and July 1, 2004 to June 30, 2007)

2. A Health Care Allowance for members of the Pension Plan as of July 1, 2004 who subsequently elect early retirement under the Pension Plan and who have 10 or more years of continuous service with the University on their early retirement date was implemented. The University will provide a health care account between their date of retirement and age 65 under the terms and conditions below. This non-bankable fund

will be used to reimburse eligible retirees and their spouses for eligible health-related expenses, including health care premiums, as defined in the *Income Tax Act.* 

The health care account will be created annually for each eligible retiree for the period from July 1st to the following June 30th (the "plan year"), in the following amounts:

- (i) For coverage for a full plan year, the amount will be \$500.
- (ii) For a member who retires in the first 6 months of the plan year, the amount for that plan year will be \$500. For a member who retires in the second 6 months of the plan year, the amount for that plan year will be \$250.
- (iii) In the plan year in which an eligible retiree attains age 65, if the retiree attains age 65 in the first 6 months of the plan year, the amount for that plan year will be \$250. If the retiree attains age 65 in the second 6 months of the plan year, the amount for that plan year will be \$500.

The health care account will cease on the earlier of the retiree's death or the end of the month in which the retiree attains age 65.

The health care account will be extended under the terms and conditions above to existing retirees under the Pension Plan who retired early after June 30, 1998 (i.e., not eligible for the health care account available to those in receipt of a pension on June 30, 1998), starting July 1, 2004.

Deferred pensioners who commence their pension before age 65 are ineligible for this health care account.

Reimbursement under the health care account program in place prior to this agreement and under this new health care account program will be made from the funds set aside in the stabilization fund. This new health care account program will cease once all the funds in the stabilization fund have been exhausted. Starting July 1, 2004, the University will credit interest to the stabilization fund on an annual basis at the rate earned on the University's cash account.

(Source Document: Appendix X, Memorandum of Agreement Re: Pension Plan, Collective Agreement between Trent University and OPSEU, July 1, 2004 to June 30, 2007)

# APPENDIX II- Student Employees

# MEMORANDUM OF UNDERSTANDING

## Re: Student Employees

The University and the Union hereby understand and agree that:

- 1. Students, including graduate students, employed by the University, are excluded from the bargaining unit as per the recognition clause of this Collective Agreement (4.1).
- 2. An OPSEU bargaining unit position shall not be reduced in hours or be declared redundant in order to employ Student employees.
- 3. Students excluded from the bargaining unit, may be employed to support, but not to replace bargaining unit positions.
- 4. Students may be employed in an OPSEU position within the scope of the bargaining unit and in accordance with Article 10.1.
- 5. The terms of employment of Student Employees is conditional upon all other relevant policies, agreements, and programs (i.e. Graduate Student Status, Trent Work Study Program, other Collective Agreements, etc).
- 6. It is understood that as part of the University's Strategic Mandate Agreement (SMA) and current government directives, there will be an expansion in experiential learning programs at Trent University. While it is understood that this will likely result in additional student experiential learning positions at Trent University, the intention is not to replace bargaining unit positions. The Employer agrees to provide periodic updates to the union at Joint Committee regarding changes in experiential learning at Trent.
- 7. Entirely without prejudice to the University's position that students, including graduate students, are excluded from the Collective Agreement, the University will provide the Union with tracking and quantification of student employees. This will include those employed and placed at Trent University in positions associated with experiential learning programs. Reports will be provided on a quarterly basis, or as requested by the Union, and shall include, if technically feasible: job descriptions, pay rates, duration of the contract/ position, department and location of the position and supervisor of the position.

# **APPENDIX III- Pension Plan**

## Memorandum of Agreement

### Re: Pension Plan and Retiree Health Care Reimbursements

### 1. Reduced Early Retirement Provision

For retirement during and after the term of this Agreement under the Pension Plan's reduced early retirement provision, the following reductions will apply:

- (i) For active members of the Pension Plan who as of July 1, 2007 are within 10 years of their normal retirement date, 4% per year from the earlier of a member's unreduced early retirement date and normal retirement date.
- (ii) For all other active members, 6% per year from the earlier of a member's unreduced early retirement date and normal retirement date.
- 2. Unreduced Early Retirement Provision

Active members of the pension plan who as of July 1, 2017 are within 10 years of their normal retirement date will be entitled to an unreduced pension on the first day of the month following the date the member has both attained age 60 and 80 points, where points is determined as age plus continuous service, both calculated in years plus completed months.

Active members of the pension plan who as of July 1, 2017 are not within 10 years of their normal retirement date will be entitled to an unreduced pension on the first day of the month following the date the member has both attained age 62 and 80 points, where points is determined as age plus continuous service, both calculated in years plus completed months.

#### 3. Indexation Provisions

For members who retire or terminate on or after July 2, 2007, the indexation provisions will be as follows:

- Pension benefits will be increased as of each July 1st by the amount of indexation that can be provided by the Cumulative Excess Investment Earnings of the pension fund, where Cumulative Excess Investment Earnings are defined as in (ii) below.
- (ii) Cumulative Excess Investment Earnings are equal to the percentage value that is the cumulative excess/deficiency of the average annualized rate of return on the pension fund (on a market value basis), net of fund expenses, above/below 6.50% for the applicable period, as described below, ending on the immediately preceding March 31st. The starting measurement date for the rate of return and the cumulative excess/deficiency will be April 1, 2007. For each of the 12month periods starting April 1st in the period from April 1, 2007 to March 31, 2010, the rate of return used in the calculation will be for the 12-month period ending on the immediately preceding March 31st and thereafter the rate of

return used in the calculation will be determined over the four-year period ending on the immediately preceding March 31st.

- (iii) Increases to pensions in payment as of July 1st will not exceed 50% of the annual increase in the Consumer Price Index determined as of the immediately preceding March 31st. For increases to deferred vested pensions, the increase will not exceed 50% of the lesser of the increase in the Consumer Price Index and the increase in the Average Industrial Wage, both determined as of the preceding March 31st.
- (iv) Where the Cumulative Excess Investment Earnings are more than sufficient to provide the above indexation in a Plan Year (as defined under the Pension Plan), the excess will be used to provide increases in respect of any one or more of the previous five Plan Years, up to the maximum amounts shown in (iii), applied in reverse chronological order.
- (v) The percentage expended on indexation in (iii) and (iv) above will be deducted from the Cumulative Excess Investment Earnings.
- (vi) If the Cumulative Excess Investment Earnings are negative, there will be no indexation; however, there will be no reduction in pension benefits.
- (vii) To the extent that the surplus as of a valuation date exceeds a contingency reserve, the surplus in excess of the contingency reserve will be available for further surplus-based indexation subject to the following:
  - The contingency reserve shall be 5% of the Pension Plan liabilities.
  - The Pension Plan liabilities for purposes of determining the surplus and contingency reserve shall include the indexing reserve included in the actuarial valuation.
  - Any surplus-based indexation shall not result in the surplus decreasing below the level of the contingency reserve or shall not create a solvency deficiency.
  - No surplus-based indexation will be granted if there is a solvency deficiency as of the valuation date.

For clarity, the existing indexation provisions under the Pension Plan will continue to apply to members who retired or terminated up to and including July 1, 2007.

4. Member Required Contributions

Member required contributions will increase from 8.5% of Earnings to 9% of Earnings effective July 1, 2015.

5. University Contributions

Over the period from July 1, 2015 to the effective date of the first filed actuarial valuation that certifies the Pension Plan is fully funded on both a going concern and solvency basis, such funded status determined without any asset smoothing, the University's total contributions to the Pension Plan on a cumulative basis will not be less than an amount calculated over the same period as the sum of 9% of the annual participant salary base and an annual special payment of \$1,126,308. For clarity, the employer may make a lower contribution in a given year in the event that they have contributed more than their minimum in a prior year.

Once the plan is fully funded on both a going concern and solvency basis, the parties agree that the pension subcommittee will review past and current member and employer funding and past benefit changes that have been made.

6. Credits For Special Payments

Any special payments made by the University in a Plan Year in connection with a going concern unfunded liability and/or solvency deficiency can be used as a credit against University contributions in any of the following 10 Plan Years provided that the use of any such credit does not create or increase a going concern unfunded liability and/or solvency deficiency.

7. Minimum Termination Benefit of Two Times Contributions Plus Interest

Effective July 1, 2015 the Pension Plan will be amended to remove the minimum termination benefit of two times the accumulated Member's Required Contributions plus interest in respect of such contributions made on and after July 1, 2015.

8. Pension Subcommittee meetings

While the pension plan is not fully funded, and beyond if the Pension subcommittee so decides is necessary or prudent, the Pension Subcommittee will meet semi-annually, with some of the consultants attending by teleconference if necessary, to discuss returns, the state of the fund etc.

9. Health Care Reimbursements

For members of the Pension Plan as of July 1, 2007 who subsequently elect early retirement under the Pension Plan and who have 10 or more years of continuous service with the University on their early retirement date, the University will provide a health care account between their date of retirement and age 65 under the terms and conditions below:

(i) This non-bankable fund will be used to reimburse eligible retirees and their spouse for eligible health-related expenses, including health care premiums, as defined under the Income Tax Act. Reimbursements will be made from the funds set aside in the Stabilization Fund, unless there are not sufficient funds in the Stabilization Fund, in which case, reimbursement will be made by the University.

- (ii) The health care account will be created annually for each eligible retiree for the period from July 1st to the following June 30th (the "plan year") in the amount of \$750 for a retiree without a spouse at the beginning of the plan year and \$1,500 for a retiree with a spouse at the beginning of the plan year, subject to the adjustments in (iii) and (iv) below for partial years.
- (iii) For a member who retires in the second six months of the plan year, the amount for that plan year will be one half of the amount in (ii).
- (iv) In a plan year in which an eligible retiree attains age 65, if the retiree attains age 65 in the first six months of the plan year, the amount for that plan year will be one half of the amount in (ii). If the retiree attains age 65 in the second six months of the plan year, the amount for that plan year will be the amount in (ii).
- (v) The health care account will cease at the earlier of the retiree's death or the end of the month in which the retiree attains age 65.
- (vi) Deferred pensioners who commence their pension before age 65 are ineligible for this health care account.

For clarity, for retirees up to and including July 1, 2007, the health care reimbursements will continue to be made in accordance with the Agreement in effect up to June 30, 2007.

# **APPENDIX IV- Restructuring**

# LETTER OF UNDERSTANDING

## Re: Restructuring

The Employer shall consult with the Union regarding any restructuring which directly affects OPSEU positions insofar as practical and reasonable in advance of advising staff of any anticipated change.

Signed at Peterborough this 1st day of June 2004.

# **APPENDIX V- RLC – Essential Workers**

# LETTER OF UNDERSTANDING

## Re: Residence Life Coordinators – Essential Workers

Given that the position of Residence Life Coordinator is a live-in position that requires the incumbent to be on-call and available 24 hours on a rotational basis to be responsible for students in residence, they are to be considered as essential workers in the event of an OPSEU labour dispute.

For the duration of the labour dispute, these employees will perform only the duties of their OPSEU positions as defined by the parties prior to a labour dispute.

Signed at Peterborough this 22nd day of June 2012.

# **APPENDIX VI- LTD Benefits**

# LETTER OF UNDERSTANDING

# RE: LTD Benefits

The parties agree, for the duration of this agreement only, to the following:

- 1. Effective July 1, 2012 the employer agrees to pay eighty percent (80%) of LTD premiums for OPSEU members.
- The parties agree that a Subcommittee of Joint Committee, consisting of two Union representatives and two management representatives, will meet to review, explore, and discuss issues around the OPSEU LTD plan. The subcommittee shall advise OPSEU and the Administration of any proposed recommendations in advance of the next Collective Bargaining process.
- In an effort to ensure improvements in LTD experience, the Union agrees to support the Employer's engagement of the services of an Occupational Health Nurse to assist with successful return to work planning.

Signed at Peterborough this 10 day of July 2012.

# **APPENDIX VII- Lieu Time**

## LETTER OF UNDERSTANDING

## RE: Lieu Time

The parties agree that Human Resources will provide OPSEU with an annual report of lieu time worked by members. This report will be provided to OPSEU on July 30 of each year.

Human Resources will commit to exploring the feasibility of providing quarterly reports of accrued lieu time at such time that a new HRIS system is implemented. The results of such exploration will be brought to joint committee no later than 6 months after the implementation of a new HRIS system.

Signed at Peterborough on this 22 day of June 2012.

# **APPENDIX VIII- Benefits over age 65**

# LETTER OF UNDERSTANDING

## RE: Benefits for Active OPSEU employees over age 65

- Health and Dental coverage will be extended, as per the existing OPSEU group benefit plan for those employees 65 and over so long as the employee remains employed in a regular OPSEU position for more than seventeen and one half (17.5) hours per week. Eligibility will be subject to Article 14.1.
- 2. Effective July 1, 2012, at normal retirement date, an active employee's coverage for life insurance is reduced by 50%, rounded to the next highest \$1,000, if not already a multiple of \$1,000, to a maximum of \$50,000. At age 70, an active employee's coverage ends.
- 3. LTD the Employer agrees, for active employees, to extend a modified LTD coverage to age 70. This coverage would work as follows: If an employee is disabled prior to their 65<sup>th</sup> birthday Sun Life will pay LTD benefits to age 65 or for 2 years whichever is longer. If an employee is working past age 65 and became disabled after age 65 then their LTD benefits would be payable for 2 years as long as those 2 years do not extend beyond their 70<sup>th</sup> birthday. In all cases the LTD benefits cease at the latest at age 70.

E.g. Disabled at age 61- LTD benefits to age 65 Disabled at age 64- LTD benefits for 2 years Disabled at age 66- LTD benefits for 2 years Disabled at age 69- LTD benefits to employees age 70

This would apply only to employees who became disabled after the change.

Signed at Peterborough on this 10 day of July 2012.

# **APPENDIX IX- Training**

# LETTER OF UNDERSTANDING

## **RE:** Training Initiatives

The parties agree that Joint Committee will meet to review, explore and recommend potential opportunities for increased professional development training for members once per year. These recommendations shall be considered in the creation of the annual Staff Development Training Schedule. The parties understand that all attempts will be made to ensure this increase in training is within existing training resources.

Signed at Peterborough on this 26 day of June 2012.

# **APPENDIX X- Backfill of Contract Positions**

Letter of Understanding

*Whereas*, Trent University recognizes the value and importance of providing staff professional development opportunities and opportunities for career growth; and

*Whereas*, the parties seek to support Managers' ability to fill contract vacancies in a timely and efficient manner; and

*Therefore*, when an OPSEU contract vacancy exists, the Parties agree to the following protocol:

- 1. Members in the Career Development program will be considered for contract vacancies prior to other applicants as per Article 19.1.
- 2. If the vacant contract position is not filled by a member in the Career Development program, other internal members will be considered for a Contract Secondment opportunity, subject to the conditions outlined below.
- 3. Posting requirements for the Contract position(s) will be as outlined in Article 18.1, unless otherwise agreed upon by the parties.
- 4. A member can apply for or be appointed to a contract Secondment at any band, subject to the provisions listed below. The contract position's band and the step closest to the current rate of pay will apply for the duration of the contract.
- 5. An internal member may be considered for a Contract Secondment when:
  - The seconded staff member meets the qualifications (core plus equivalency) of the vacant position, subject to assessment through an interview and/or testing; and
  - b. The secondment makes operational sense. In determining operational sense, the following considerations will be taken into account:
    - i. the seconded staff member is able to perform the contract work with limited training;
    - ii. The seconded staff member's vacant home position is anticipated to be able to be filled for the duration of the secondment;
    - iii. the employee must be available for the duration of the contract vacancy.
  - c. The Employer reserves the right to determine operational priorities and may decide that a staff member is not able to be released to a Contract Secondment position if there is a valid business reason that the member is required to stay in their home position.
  - d. Subject to approval of the outgoing manager(s).

- 6. Prior to offering a Contract Secondment appointment, the Employer will advise the Union of the backfill plan.
- 7. Internal applicants will be considered at the same time as external candidates.
- 8. In the event that multiple internal candidates, or an internal and an external candidate are determined to be relatively equal, as outlined in Article 18.3, seniority or internal status will be the deciding factor.
- 9. During the time that a member is in a Contract Secondment assignment, their home position will be held and may be filled by another internal member, subject to the provisions of Article 10.1 (c), Article 13.14, and the terms and conditions of this Letter of Understanding.
- The parties agree that any permanent changes affecting OPSEU positions shall be brought forward to OPSEU as a restructuring in accordance with Appendix V – Restructuring.
- 11. The parties agree to review the terms and conditions of this Letter of Understanding after a six-month trial period. At the conclusion of the six-month trial period, the parties will review the status of the agreement. At any time, the terms above may be renegotiated by the parties at Joint Committee.

Signed at Peterborough this 10<sup>th</sup> day of May 2018.

# **APPENDIX XI- Job Competition Equivalency**

Letter of Understanding

The University will include a statement on the Human Resources Website:

The University encourages all qualified applicants to apply for positions; training and/or experience may be substituted for formal academic training, at the University's discretion.

It is understood, that should an internal applicant not possess the posted educational credentials, the University's decision not to consider the candidate, is not grievable.

Signed at Peterborough this 10<sup>th</sup> day of May 2018.

# **APPENDIX XII- Survivor Income Benefit**

# MEMORANDUM OF AGREEMENT

# **Re: Survivor Income Benefit**

WHEREAS, the Survivor Income Benefit is an employee paid death benefit; and,

WHEREAS, the Survivor Income Benefit is considered a declining benefit across the industry and is not widely offered; and,

WHEREAS, the Survivor Income Benefit is a very expensive benefit for the member, and offers a very small payout relative to cost as the rates have increased dramatically over the last few years; and,

WHEREAS, only two (2) OPSEU members are currently enrolled in the Survivor Income Benefit.

THEREFORE, the parties agree:

- 1. Within 90 days of the signing of this agreement the University will communicate and/ or meet with the member and a Union representative to education the member(s) about the Survivor Income Benefit, on the costs and benefits of the existing program and other existing options available to them.
- 2. Sun Life will allow the member(s) to convert their Survivor Income Benefit to a 1X Optional Life Insurance without having to provide evidence of insurability.
- 3. If, and when, these two (2) members agree to withdraw or convert, the parties agree that the Survivor Income Benefit will cease.
- 4. The parties agree that the Survivor Income Benefit will not be offered to any new entrants, effective the date of the signing of this agreement.

Signed on this 10<sup>th</sup> day of May, 2018.

# APPENDIX XIII- Pension Plan 2018-2022

#### Memorandum of Agreement

### Re: Pension Plan

WHEREAS, in bargaining for a renewal collective agreement with OPSEU, the University has proposed modifications to the Pension Plan including the following:

#### CAAT Plan or University Sector JSPP

The University and OPSEU will pursue a merger of the OPSEU portion of the OPSEU / Exempt Pension Plan with the CAAT Plan or University Sector JSPP or other pension plan such as Ontario Pension Trust, and in the event that the Board of Governors and the OPSEU membership support such a merger, the parties will amend the existing OPSEU collective agreement for the purpose of concluding negotiations of a plan merger, recognizing that such a merger would be subject to ratification by the Board of Governors and the OPSEU membership. The University reserves the right to address the Exempt portion of the OPSEU / Exempt Pension Plan in a different manner.

A joint working group to be established to formalize a detailed project plan to address the potential requirements of merging the Pension Plan with CAAT or a University Sector JSPP. This group will be comprised of 2 OPSEU representatives, 1 Exempt and 2 University Representatives.

#### 1. Joint Information Sessions

At OPSEU's request, the University and OPSEU will offer a Joint Information session during each reading week regarding the University Sector JSPP and the status of the Pension Plan.

#### 2. Additional Meetings

In addition to the sub-committee meetings held in November or December of every year during which the financial position of the Pension Plan and the pension fund investment performance is reviewed, an additional meeting will be established in the second calendar quarter at the request of OPSEU including suggested agenda items from OPSEU.

The University will arrange for an orientation session with a representative of AON Hewitt, for any current or new OPSEU members appointed by the Local to the Pension Sub- Committee.

#### 3. Joint Committee

Pension will be a standing item on monthly Joint Committee agenda. One of these meetings will include an information session between the OPSEU and the AVP, Finance and VP Administration during budget consultation process.

#### 4. Member Required Contributions

Member required contributions will increase from 9.0% of Earnings (as defined under the Pension Plan) to 9.8% of Earnings effective June 30, 2020 and will continue to remain at this level.

Signed on this 15<sup>th</sup> day of June, 2018.

# APPENDIX XIV- Vacation pay-out for Regular Caretaking Staff

Letter of Understanding

The parties agree to the following terms and conditions:

- 1. Until such a time that the University implements an HRIS system, any vacation pay-out for extra hours worked from January to June each year will be paid in mid-July and vacation pay-out on extra hours worked from July to December each year will be paid in mid-January.
- 2. No later than 60 days prior to the development of the vacation module being developed for the HRIS, the University and OPSEU will meet to discuss when and how vacation will be paid out, or accrued for usage in the following year, for additional hours for caretaking staff subject to the capability and feasibility of the same in the HRIS.

Signed on this 10<sup>th</sup> day of May, 2018.

# **APPENDIX XV- Vacation for Recurring Employees**

The parties agree to the following terms and conditions:

- 1. Subject to Article 15.1 of the Collective Agreement, recurring employees shall be eligible to take up to half of their vacation entitlement between their return from layoff and January 1.
- 2. The second half of the vacation entitlement, in addition to any time not taken between their return and January 1 may be taken between January 2 and their lay off date.
- 3. To ensure over use does not occur, vacation entitlements will be calculated based on an employee's hours as outlined in their appointment letter. For clarity, additional hours, and hours worked during the summer are not to be included in the calculation for vacation time.
- 4. Unused vacation entitlement, additional hours, and hours worked during the summer shall be paid out in May or September as elected by the employee.
- 5. Should an employee's employment with Trent University end, and the employee has taken vacation time in excess of what they are eligible at that time, the employee will have the monies deducted from their final pay.

Signed on this 10<sup>th</sup> day of May, 2018.

#### SALARY GRID - YEAR I - July 1, 2018 to June 30, 2019 (Based on 35 hours per week)

Note: Initial calculations are based on Monthly Salary figures. Annual Salary = Monthly x 12;

BAND	POINT	STEP	STEP	STEP	STEP	STEP	STEP	STEP		
#	RANGE	1	2	3	4	5	6	7		
2% increa	ase		Annual Salary (Based on 35 hours per week)							
		1	2	3	4	5	6	7		
1	-131	32,010	32,907	33,831	34,781	35,756	36,760	37,792		
2	132-163	34,781	35,756	36,760	37,792	38,852	39,943	41,063		
3	164-195	37,792	38,852	39,943	41,063	42,215	43,400	44,618		
4	196-227	41,063	42,215	43,400	44,618	45,870	47,157	48,481		
5	228-259	44,618	45,870	47,157	48,481	49,841	51,240	52,678		
6	260-291	48,481	49,841	51,240	52,678	54,156	55,676	57,238		
7	292-323	52,678	54,156	55,676	57,238	58,845	60,496	62,194		
8	324-355	57,238	58,845	60,496	62,194	63,939	65,733	67,578		
9	356-387	62,194	63,939	65,733	67,578	69,474	71,424	73,428		
10	388-419	67,578	69,474	71,424	73,428	75,488	77,607	79,785		
11	420-451	73,428	75,488	77,607	79,785	82,024	84,326	86,692		
12	452-483	79,785	82,024	84,326	86,692	89,125	91,626	94,197		
13	484-515	86,692	89,125	91,626	94,197	96,840	99,558	102,352		
			Мс	onthly Salary (I	Based on 35 h	ours per week	)			
		1	2	3	4	5	6	7		
1	-131	2,667.46	2,742.28	2,819.27	2,898.38	2,979.71	3,063.33	3,149.30		
2	132-163	2,898.38	2,979.71	3,063.33	3,149.30	3,237.65	3,328.55	3,421.93		
3	164-195	3,149.30	3,237.65	3,328.55	3,421.93	3,517.95	3,616.66	3,718.16		
4	196-227	3,421.93	3,517.95	3,616.66	3,718.16	3,822.51	3,929.77	4,040.07		
5	228-259	3,718.16	3,822.51	3,929.77	4,040.07	4,153.43	4,270.00	4,389.80		
6	260-291	4,040.07	4,153.43	4,270.00	4,389.80	4,513.00	4,639.65	4,769.85		
7	292-323	4,389.80	4,513.00	4,639.65	4,769.85	4,903.71	5,041.33	5,182.82		
8	324-355	4,769.85	4,903.71	5,041.33	5,182.82	5,328.22	5,477.75	5,631.47		
9	356-387	5,182.82	5,328.22	5,477.75	5,631.47	5,789.50	5,952.00	6,119.01		
10	388-419	5,631.47	5,789.50	5,952.00	6,119.01	6,290.71	6,467.23	6,648.74		
11	420-451	6,119.01	6,290.71	6,467.23	6,648.74	6,835.35	7,027.13	7,224.35		
12	452-483	6,648.74	6,835.35	7,027.13	7,224.35	7,427.04	7,635.50	7,849.75		
13	484-515	7,224.35	7,427.04	7,635.50	7,849.75	8,070.03	8,296.50	8,529.30		
					Hourly Rate					
		1	2	3	4	5	6	7		
1	-131	17.59	18.08	18.59	19.11	19.65	20.20	20.76		
2	132-163	19.11	19.65	20.20	20.76	21.35	21.95	22.56		
3	164-195	20.76	21.35	21.95	22.56	23.20	23.85	24.52		
4	196-227	22.56	23.20	23.85	24.52	25.20	25.91	26.64		
5	228-259	24.52	25.20	25.91	26.64	27.39	28.15	28.94		
6	260-291	26.64	27.39	28.15	28.94	29.76	30.59	31.45		
7	292-323	28.94	29.76	30.59	31.45	32.33	33.24	34.17		
8	324-355	31.45	32.33	33.24	34.17	35.13	36.12	37.13		
9	356-387	34.17	35.13	36.12	37.13	38.17	39.24	40.35		
10	388-419	37.13	38.17	39.24	40.35	41.48	42.64	43.84		
11	420-451	40.35	41.48	42.64	43.84	45.07	46.33	47.63		
12	452-483	43.84	45.07	46.33	47.63	48.97	50.34	51.76		
13	484-515	47.63	48.97	50.34	51.76	53.21	54.70	56.24		

#### SALARY GRID - YEAR II - July 1, 2019 to June 30, 2020 (Based on 35 hours per week)

Note: Initial calculations are based on Monthly Salary figures. Annual Salary = Monthly x 12;

BAND	POINT	STEP	STEP	STEP	STEP	STEP	STEP	STEP
#	RANGE	1	2	3	4	5	6	7
1.9% increase			Ar	nnual Salary (E	Based on 35 h	ours per week)		
		1	2	3	4	5	6	7
1	-131	32,618	33,533	34,474	35,441	36,436	37,458	38,510
2	132-163	35,441	36,436	37,458	38,510	39,590	40,701	41,843
3	164-195	38,510	39,590	40,701	41,843	43,018	44,225	45,466
4	196-227	41,843	43,018	44,225	45,466	46,742	48,053	49,402
5	228-259	45,466	46,742	48,053	49,402	50,788	52,214	53,679
6	260-291	49,402	50,788	52,214	53,679	55,185	56,734	58,326
7	292-323	53,679	55,185	56,734	58,326	59,963	61,645	63,375
8	324-355	58,326	59,963	61,645	63,375	65,154	66,982	68,862
9	356-387	63,375	65,154	66,982	68,862	70,794	72,781	74,823
10	388-419	68,862	70,794	72,781	74,823	76,923	79,081	81,301
11	420-451	74,823	76,923	79,081	81,301	83,583	85,928	88,339
12	452-483	81,301	83,583	85,928	88,339	90,818	93,367	95,987
13	484-515	88,339	90,818	93,367	95,987	98,680	101,450	104,296
			Мс	onthly Salary (	Based on 35 h	ours per week	)	
		1	2	3	4	5	6	7
1	-131	2,718.14	2,794.39	2,872.84	2,953.45	3,036.32	3,121.53	3,209.14
2	132-163	2,953.45	3,036.32	3,121.53	3,209.14	3,299.17	3,391.79	3,486.94
3	164-195	3,209.14	3,299.17	3,391.79	3,486.94	3,584.79	3,685.38	3,788.81
4	196-227	3,486.94	3,584.79	3,685.38	3,788.81	3,895.13	4,004.44	4,116.83
5	228-259	3,788.81	3,895.13	4,004.44	4,116.83	4,232.34	4,351.13	4,473.21
6	260-291	4,116.83	4,232.34	4,351.13	4,473.21	4,598.74	4,727.81	4,860.48
7	292-323	4,473.21	4,598.74	4,727.81	4,860.48	4,996.88	5,137.12	5,281.29
8	324-355	4,860.48	4,996.88	5,137.12	5,281.29	5,429.46	5,581.83	5,738.46
9	356-387	5,281.29	5,429.46	5,581.83	5,738.46	5,899.50	6,065.09	6,235.27
10	388-419	5,738.46	5,899.50	6,065.09	6,235.27	6,410.23	6,590.10	6,775.07
11	420-451	6,235.27	6,410.23	6,590.10	6,775.07	6,965.22	7,160.65	7,361.61
12	452-483	6,775.07	6,965.22	7,160.65	7,361.61	7,568.16	7,780.57	7,998.90
13	484-515	7,361.61	7,568.16	7,780.57	7,998.90	8,223.36	8,454.13	8,691.36
					Hourly Rate			
		1	2	3	4	5	6	7
1	-131	17.92	18.42	18.94	19.47	20.02	20.58	21.16
2	132-163	19.47	20.02	20.58	21.16	21.75	22.36	22.99
3	164-195	21.16	21.75	22.36	22.99	23.64	24.30	24.98
4	196-227	22.99	23.64	24.30	24.98	25.68	26.40	27.14
5	228-259	24.98	25.68	26.40	27.14	27.91	28.69	29.49
6	260-291	27.14	27.91	28.69	29.49	30.32	31.17	32.05
7	292-323	29.49	30.32	31.17	32.05	32.95	33.87	34.82
8	324-355	32.05	32.95	33.87	34.82	35.80	36.80	37.84
9	356-387	34.82	35.80	36.80	37.84	38.90	39.99	41.11
10	388-419	37.84	38.90	39.99	41.11	42.27	43.45	44.67
11	420-451	41.11	42.27	43.45	44.67	45.92	47.21	48.54
12	452-483	44.67	45.92	47.21	48.54	49.90	51.30	52.74
13	484-515	48.54	49.90	51.30	52.74	54.22	55.74	57.31

#### SALARY GRID - YEAR III - July 1, 2020 to June 30, 2021 (Based on 35 hours per week)

Note: Initial calculations are based on Monthly Salary figures. Annual Salary = Monthly x 12;

BAND	POINT	STEP	STEP	STEP	STEP	STEP	STEP	STEP	
#	RANGE	1	2	3	4	5	6	7	
1.9% increase			Annual Salary (Based on 35 hours per week)						
		1	2	3	4	5	6	7	
1	-131	33,237	34,170	35,129	36,115	37,128	38,170	39,241	
2	132-163	36,115	37,128	38,170	39,241	40,342	41,475	42,638	
3	164-195	39,241	40,342	41,475	42,638	43,835	45,065	46,330	
4	196-227	42,638	43,835	45,065	46,330	47,630	48,966	50,341	
5	228-259	46,330	47,630	48,966	50,341	51,753	53,206	54,698	
6	260-291	50,341	51,753	53,206	54,698	56,233	57,812	59,434	
7	292-323	54,698	56,233	57,812	59,434	61,102	62,817	64,580	
8	324-355	59,434	61,102	62,817	64,580	66,391	68,255	70,170	
9	356-387	64,580	66,391	68,255	70,170	72,139	74,164	76,245	
10	388-419	70,170	72,139	74,164	76,245	78,384	80,584	82,846	
11	420-451	76,245	78,384	80,584	82,846	85,171	87,560	90,018	
12	452-483	82,846	85,171	87,560	90,018	92,543	95,141	97,811	
13	484-515	90,018	92,543	95,141	97,811	100,555	103,377	106,278	
			Мс	onthly Salary (	Based on 35 h	ours per week	)		
		1	2	3	4	5	6	7	
1	-131	2,769.79	2,847.48	2,927.42	3,009.56	3,094.01	3,180.84	3,270.12	
2	132-163	3,009.56	3,094.01	3,180.84	3,270.12	3,361.85	3,456.23	3,553.19	
3	164-195	3,270.12	3,361.85	3,456.23	3,553.19	3,652.90	3,755.40	3,860.79	
4	196-227	3,553.19	3,652.90	3,755.40	3,860.79	3,969.14	4,080.52	4,195.05	
5	228-259	3,860.79	3,969.14	4,080.52	4,195.05	4,312.76	4,433.81	4,558.20	
6	260-291	4,195.05	4,312.76	4,433.81	4,558.20	4,686.12	4,817.64	4,952.83	
7	292-323	4,558.20	4,686.12	4,817.64	4,952.83	5,091.82	5,234.72	5,381.63	
8	324-355	4,952.83	5,091.82	5,234.72	5,381.63	5,532.62	5,687.88	5,847.50	
9	356-387	5,381.63	5,532.62	5,687.88	5,847.50	6,011.59	6,180.32	6,353.74	
10	388-419	5,847.50	6,011.59	6,180.32	6,353.74	6,532.02	6,715.32	6,903.79	
11	420-451	6,353.74	6,532.02	6,715.32	6,903.79	7,097.56	7,296.70	7,501.48	
12	452-483	6,903.79	7,097.56	7,296.70	7,501.48	7,711.95	7,928.40	8,150.88	
13	484-515	7,501.48	7,711.95	7,928.40	8,150.88	8,379.60	8,614.76	8,856.49	
					Hourly Rate				
		1	2	3	4	5	6	7	
1	-131	18.26	18.77	19.30	19.84	20.40	20.97	21.56	
2	132-163	19.84	20.40	20.97	21.56	22.17	22.79	23.43	
3	164-195	21.56	22.17	22.79	23.43	24.09	24.76	25.46	
4	196-227	23.43	24.09	24.76	25.46	26.17	26.90	27.66	
5	228-259	25.46	26.17	26.90	27.66	28.44	29.23	30.05	
6	260-291	27.66	28.44	29.23	30.05	30.90	31.76	32.66	
7	292-323	30.05	30.90	31.76	32.66	33.57	34.51	35.48	
8	324-355	32.66	33.57	34.51	35.48	36.48	37.50	38.55	
9	356-387	35.48	36.48	37.50	38.55	39.64	40.75	41.89	
10	388-419	38.55	39.64	40.75	41.89	43.07	44.28	45.52	
11	420-451	41.89	43.07	44.28	45.52	46.80	48.11	49.46	
12	452-483	45.52	46.80	48.11	49.46	50.85	52.28	53.74	
13	484-515	49.46	50.85	52.28	53.74	55.25	56.80	58.39	

#### SALARY GRID - YEAR IV - July 1, 2021 to June 30, 2022 (Based on 35 hours per week)

Note: Initial calculations are based on Monthly Salary figures. Annual Salary = Monthly x 12;

BAND	POINT	STEP	STEP	STEP	STEP	STEP	STEP	STEP
#	RANGE	1	2	3	4	5	6	7
1.9% incr	ease		Annual Salary (Based on 35 hours per week)					
		1	2	3	4	5	6	7
1	-131	33,869	34,819	35,796	36,801	37,834	38,895	39,987
2	132-163	36,801	37,834	38,895	39,987	41,109	42,263	43,448
3	164-195	39,987	41,109	42,263	43,448	44,668	45,921	47,210
4	196-227	43,448	44,668	45,921	47,210	48,535	49,897	51,297
5	228-259	47,210	48,535	49,897	51,297	52,736	54,217	55,738
6	260-291	51,297	52,736	54,217	55,738	57,302	58,910	60,563
7	292-323	55,738	57,302	58,910	60,563	62,263	64,010	65,807
8	324-355	60,563	62,263	64,010	65,807	67,653	69,551	71,503
9	356-387	65,807	67,653	69,551	71,503	73,510	75,573	77,694
10	388-419	71,503	73,510	75,573	77,694	79,874	82,115	84,420
11	420-451	77,694	79,874	82,115	84,420	86,789	89,224	91,728
12	452-483	84,420	86,789	89,224	91,728	94,302	96,949	99,669
13	484-515	91,728	94,302	96,949	99,669	102,466	105,341	108,297
			Мо	onthly Salary (	Based on 35 h	ours per week	)	
		1	2	3	4	5	6	7
1	-131	2,822.41	2,901.58	2,983.04	3,066.74	3,152.80	3,241.27	3,332.25
2	132-163	3,066.74	3,152.80	3,241.27	3,332.25	3,425.73	3,521.90	3,620.70
3	164-195	3,332.25	3,425.73	3,521.90	3,620.70	3,722.31	3,826.76	3,934.15
4	196-227	3,620.70	3,722.31	3,826.76	3,934.15	4,044.55	4,158.05	4,274.76
5	228-259	3,934.15	4,044.55	4,158.05	4,274.76	4,394.70	4,518.05	4,644.81
6	260-291	4,274.76	4,394.70	4,518.05	4,644.81	4,775.16	4,909.17	5,046.93
7	292-323	4,644.81	4,775.16	4,909.17	5,046.93	5,188.57	5,334.18	5,483.88
8	324-355	5,046.93	5,188.57	5,334.18	5,483.88	5,637.74	5,795.95	5,958.60
9	356-387	5,483.88	5,637.74	5,795.95	5,958.60	6,125.82	6,297.75	6,474.46
10	388-419	5,958.60	6,125.82	6,297.75	6,474.46	6,656.13	6,842.91	7,034.96
11	420-451	6,474.46	6,656.13	6,842.91	7,034.96	7,232.42	7,435.34	7,644.01
12	452-483	7,034.96	7,232.42	7,435.34	7,644.01	7,858.48	8,079.04	8,305.75
13	484-515	7,644.01	7,858.48	8,079.04	8,305.75	8,538.81	8,778.44	9,024.77
					Hourly Rate			
		1	2	3	4	5	6	7
1	-131	18.61	19.13	19.67	20.22	20.79	21.37	21.97
2	132-163	20.22	20.79	21.37	21.97	22.59	23.22	23.87
3	164-195	21.97	22.59	23.22	23.87	24.54	25.23	25.94
4	196-227	23.87	24.54	25.23	25.94	26.67	27.42	28.19
5	228-259	25.94	26.67	27.42	28.19	28.98	29.79	30.63
6	260-291	28.19	28.98	29.79	30.63	31.48	32.37	33.28
7	292-323	30.63	31.48	32.37	33.28	34.21	35.17	36.16
8	324-355	33.28	34.21	35.17	36.16	37.17	38.22	39.29
9	356-387	36.16	37.17	38.22	39.29	40.39	41.52	42.69
10	388-419	39.29	40.39	41.52	42.69	43.89	45.12	46.38
11	420-451	42.69	43.89	45.12	46.38	47.69	49.02	50.40
12	452-483	46.38	47.69	49.02	50.40	51.81	53.27	54.76
13	484-515	50.40	51.81	53.27	54.76	56.30	57.88	59.50

# SCHEDULE B

Please refer to <u>https://www.trentu.ca/humanresources/human-resources-services/job-evaluation/job-descriptions</u> for a complete listing of positions with banding and job classification and number.

#### THIS COLLECTIVE AGREEMENT

### SIGNED AT PETERBOROUGH, ONTARIO

THIS H DAY OF JULY 2018

For the University:

L. Groarke President and Vice-Chancellor

S. Pillar Vice President (Administration)

D. Large Chief Spokesperson

S. Williams AVP, Human Resources

T. Al-idrissi Negotiating Committee

end Tuck

C. Turk Negotiating Committee

J. Coulter Negotiating Committee

Licte Mrkeen

N. McKeen Negotiating Committee

For the Union:

Wan Han

W. Thomas OPSEU President

C. Johnston

OPSEU Negotiator

A. Devlin President, Local 365

D. Howard-Gill Negotiating Committee

C. Williams Negotiating Committee

M. Keefer

Negotiating Committee

00

D. Earle Negotiating Committee